

Cochrane Database of Systematic Reviews

Amantadine and rimantadine for influenza A in adults (Review)

Jefferson T, Demicheli V, Di Pietrantonj C, Rivetti D

Jefferson T, Demicheli V, Di Pietrantonj C, Rivetti D. Amantadine and rimantadine for influenza A in adults. *Cochrane Database of Systematic Reviews* 2006, Issue 2. Art. No.: CD001169. DOI: 10.1002/14651858.CD001169.pub3.

www.cochranelibrary.com



TABLE OF CONTENTS

| HEADER | |
|--|-------------------------------------|
| ABSTRACT | |
| PLAIN LANGUAGE SUMMARY | |
| BACKGROUND | |
| DBJECTIVES | |
| METHODS | |
| RESULTS | |
| DISCUSSION | |
| AUTHORS' CONCLUSIONS | |
| CKNOWLEDGEMENTS | |
| REFERENCES | |
| HARACTERISTICS OF STUDIES | |
| DATA AND ANALYSES | |
| Analysis 1.1. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 1 Influe | nza cases |
| Analysis 1.2. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 2 ILI cas | ses |
| Analysis 1.3. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 3 Adver | se effects |
| Analysis 1.4. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 4 Viral upper airways at 2 to 5 days. | |
| Analysis 1.5. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 5 Influer | |
| Analysis 2.1. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 1 Influe | nza cases. |
| Analysis 2.2. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 2 ILI cas | ses |
| Analysis 2.3. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 3 Adver | se effects. |
| Analysis 2.4. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 4 Influe | nza cases (asymptomatic) |
| Analysis 3.1. Comparison 3 Oral amantadine versus oral rimantadine (prophylaxis), Outcome | e 1 Influenza cases |
| Analysis 3.2. Comparison 3 Oral amantadine versus oral rimantadine (prophylaxis), Outcome | e 2 ILI cases. |
| Analysis 3.3. Comparison 3 Oral amantadine versus oral rimantadine (prophylaxis), Outcome | e 3 Adverse effects. |
| Analysis 4.1. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 1 Duratio or more) in days. | on of fever (37 degrees centigrade |
| Analysis 4.2. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 2 Cases w | |
| Analysis 4.3. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 3 Adverse | |
| Analysis 4.4. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 4 Duration | |
| Analysis 4.5. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 5 Viral nasa airways at 2 to 5 days. | |
| Analysis 5.1. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 1 Duratio or more) in days. | |
| Analysis 5.2. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 2 Cases w | vith fever at 48 hours |
| Analysis 5.3. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 3 Adverse | |
| Analysis 5.4. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 4 Viral upper airways at 2 to 5 days. | |
| Analysis 6.1. Comparison 6 Oral amantadine versus oral rimantadine (treatment), Outcome centigrade or more) in days. | |
| Analysis 6.2. Comparison 6 Oral amantadine versus oral rimantadine (treatment), Outcome 2 | |
| Analysis 6.3. Comparison 6 Oral amantadine versus oral rimantadine (treatment), Outcome 3 | |
| Analysis 7.1. Comparison 7 Oral or inhaled amantadine versus placebo or aspirin, Outcome 1 V in upper airways at 2 to 5 days. | 'iral nasal shedding or persistence |
| Analysis 8.1. Comparison 8 Oral amantadine versus standard medication (treatment), Outcon centigrade or more) in days. | ne 1 Duration of fever (37 degrees |
| Analysis 8.2. Comparison 8 Oral amantadine versus standard medication (treatment), Outcom | |
| Analysis 8.3. Comparison 8 Oral amantadine versus standard medication (treatment), Ou persistence in upper airways at 2 to 5 days. | tcome 3 Viral nasal shedding or |
| Analysis 9.1. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 1 Syn (respiratory illness). | nptoms score in confirmed cases |
| | |



| Analysis 9.2. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 2 Symptoms score in confirmed cases (constitutional illness). | 54 |
|--|----|
| Analysis 9.3. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 3 Adverse effects - local - nasal irritation. | 54 |
| Analysis 9.4. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 4 Viral nasal shedding or persistence in upper airways at 2 to 5 days. | 55 |
| ADDITIONAL TABLES | 55 |
| FEEDBACK | 55 |
| WHAT'S NEW | 56 |
| HISTORY | 56 |
| CONTRIBUTIONS OF AUTHORS | 56 |
| DECLARATIONS OF INTEREST | 57 |
| SOURCES OF SUPPORT | 57 |
| NOTES | 57 |
| INDEX TERMS | 57 |



[Intervention Review]

Amantadine and rimantadine for influenza A in adults

Tom Jefferson¹, Vittorio Demicheli², Carlo Di Pietrantonj², Daniela Rivetti³

¹The Cochrane Collaboration, Roma, Italy. ²Servizio Regionale di Riferimento per l'Epidemiologia, SSEpi-SeREMI - Cochrane Vaccines Field, Azienda Sanitaria Locale ASL AL, Alessandria, Italy. ³Public Health Department, Servizio di Igiene e Sanita' Pubblica, ASL 19 Asti, Asti, Italy

Contact address: Tom Jefferson, The Cochrane Collaboration, Via Puglie 23, Roma, 00187, Italy. jefferson.tom@gmail.com, jefferson@assr.it.

Editorial group: Cochrane Acute Respiratory Infections Group **Publication status and date:** Stable (no update expected for reasons given in 'What's new'), published in Issue 7, 2012.

Citation: Jefferson T, Demicheli V, Di Pietrantonj C, Rivetti D. Amantadine and rimantadine for influenza A in adults. *Cochrane Database of Systematic Reviews* 2006, Issue 2. Art. No.: CD001169. DOI: 10.1002/14651858.CD001169.pub3.

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Amantadine hydrochloride (amantadine) and rimantadine hydrochloride (rimantadine) have antiviral properties, but they are not widely used due to a lack of knowledge of their potential value and concerns about possible adverse effects.

This review was first published in 1999 and updated for the fourth time in April 2008.

Objectives

The objective of this review was to assess the efficacy, effectiveness and safety ('effects') of amantadine and rimantadine in healthy adults.

Search methods

We searched the Cochrane Central Register of Controlled Trials (CENTRAL) (*The Cochrane Library* 2008, issue 1), MEDLINE (1966 to April Week 4, 2008), EMBASE (1990 to April 2008) and reference lists of articles.

Selection criteria

Randomised and quasi-randomised studies comparing amantadine and/or rimantadine with placebo, control medication or no intervention, or comparing doses or schedules of amantadine and/or rimantadine in healthy adults.

Data collection and analysis

For prophylaxis (prevention) trials we analysed the numbers of participants with clinical influenza (influenza-like-illness or ILI) or with confirmed influenza A and adverse effects. Analysis for treatment trials was of the mean duration of fever, length of hospital stay and adverse effects.

Main results

Amantadine prevented 25% of ILI cases (95% confidence interval (CI) 13% to 36%), and 61% of influenza A cases (95% CI 35% to 76%). Amantadine reduced duration of fever by one day (95% CI 0.7 to 1.2). Rimantadine demonstrated comparable effectiveness, but there were fewer trials and the results for prophylaxis were not statistically significant. Both amantadine and rimantadine induced significant gastrointestinal (GI) adverse effects. Adverse effects of the central nervous system and study withdrawals were significantly more common with amantadine than rimantadine. Neither drug affected the rate of viral shedding from the nose or the course of asymptomatic influenza.



Authors' conclusions

Amantadine and rimantadine have comparable efficacy and effectiveness in relieving or treating symptoms of influenza A in healthy adults, although rimantadine induces fewer adverse effects than amantadine. The effectiveness of both drugs in interrupting transmission is probably low. Resistance of influenza viruses to amantadine is a serious worldwide problem as shown by recent virological surveillances. Both drugs have adverse gastrointestinal (stomach and gut) effects, but amantadine can also have serious effects on the nervous system. They should only be used in an emergency when all other measures fail.

PLAIN LANGUAGE SUMMARY

Antiviral drugs amantadine and rimantadine for preventing and treating the symptoms of influenza A in adults

The drugs amantadine and rimantadine can both help prevent and relieve the symptoms of influenza A in adults, but amantadine has more adverse effects.

The flu can be caused by many different viruses. One type is influenza A, with headaches, coughs and runny noses that can last for many days and lead to serious illnesses such as pneumonia. Amantadine and rimantadine are antiviral drugs. The review of trials found that both drugs are similarly helpful in relieving the symptoms of influenza A in adults, but only when there is a high probability that the cause of the flu is influenza A (a known epidemic, for example). It is likely that neither drug will interrupt the spread of influenza A and by treating symptoms may encourage viral spread in the community by people who are feeling better but are still infectious. Resistance of influenza viruses to amantadine is a serious worldwide problem as shown by recent surveys. Both drugs have adverse gastrointestinal (stomach and gut) effects, but amantadine can also have serious effects on the nervous system. They should only be used in an emergency when all other measures fail.



BACKGROUND

Description of the intervention

The M2 ion channel blocking antiviral compounds amantadine hydrochloride (amantadine) and rimantadine hydrochloride (rimantadine) were licensed in 1976 and 1993 respectively as anti-influenza drugs in the USA. Recently the World Health Organization has encouraged member countries to use antivirals in seasonal influenza "interpandemic periods". The rationale given is as follows: "wide scale use of antivirals and vaccines during a pandemic will depend on familiarity with their effective application during the interpandemic period. The increasing use of these modalities will expand capacity and mitigate the morbidity and mortality of annual influenza epidemics. Studies conducted during the interpandemic period can refine the strategies for use during a pandemic" (WHO 2005). It is also likely that given their low cost amantadine and rimantadine may be used in epidemic or pandemic situations.

This review was first published in 1999 and updated for the fourth time in April 2008.

OBJECTIVES

- 1. To identify, retrieve and assess all studies evaluating the effects of amantadine and rimantadine on influenza A in healthy adults.
- 2. To assess the effectiveness of amantadine and/or rimantadine in preventing cases of influenza A (prophylaxis) in healthy adults, both at an individual level and to interrupt transmission.
- 3. To assess the effectiveness of amantadine and/or rimantadine in shortening or reducing the severity of influenza A in healthy adults (treatment).
- 4. To estimate the frequency of adverse effects associated with amantadine and/or rimantadine administration in healthy adults.

METHODS

Criteria for considering studies for this review

Types of studies

Any randomised or quasi-randomised studies comparing amantadine and/or rimantadine in humans with placebo, control medication or no intervention or comparing doses or schedules of amantadine and/or rimantadine. Only studies assessing protection or treatment from exposure to naturally occurring influenza were considered initially.

Types of participants

Apparently healthy individuals (with no known pre-existing chronic pathology known to aggravate the course of influenza) aged 14 to 60.

Types of interventions

Amantadine and/or rimantadine as prophylaxis and/or treatment for influenza, irrespective of target viral antigenic configuration.

Types of outcome measures

Clinical

Numbers and/or severity (however defined) of influenza cases and/ or deaths occurring in amantadine and/or rimantadine and placebo or control groups.

Infectivity of index cases (measured by variables such as length of nasal shedding of influenza viruses or persistence in the upper airways).

Adverse effects

Number and seriousness of adverse effects, including cases of malaise, nausea, fever, arthralgias, rash, headache and more generalised and serious signs.

Search methods for identification of studies

Electronic searches

In the original review, published in *The Cochrane Library* 1999, issue 3, an electronic search of MEDLINE was carried out using the extended search strategy of the Cochrane Acute Respiratory Infections (ARI) Group (ARI Group 1998) with the following search terms or combined sets from 1966 to the end of 1997 in any language:

Influenza Route (oral) Route (parenteral) Amantadine Rimantadine

We read the bibliography of retrieved articles and of reviews of the topic in order to identify further trials. We also carried out a search of the Cochrane Controlled Trials Register (CCTR) and of EMBASE (1985 to 1997). In order to locate unpublished trials we wrote to the following:

- manufacturers;
- researchers active in the field;
- first or corresponding authors of studies evaluated (but not necessarily included) in the review.

In the first updated review published in 2001, the Cochrane Acute Respiratory Infections Group's Trials Register was searched in March 2001, and CENTRAL (*The Cochrane Library* 2001, issue 2) was also searched for new trials.

In the second updated review in 2003, we searched the Cochrane Central Register of Controlled Trials (CENTRAL) (*The Cochrane Library* 2003, issue 4), MEDLINE (January 1966 to November week 2, 2003), EMBASE (1990 to November 2003) and reference lists of articles. We also contacted manufacturers, researchers and authors. There were no language restrictions.

In the third updated review, we searched the Cochrane Central Register of Controlled Trials (CENTRAL) (*The Cochrane Library* 2005, issue 3), MEDLINE (2003 to August Week 4, 2005), EMBASE (October 2003 to July 2005) and reference lists of articles. We also contacted manufacturers, researchers and authors. There were no language restrictions.

Amantadine and rimantadine for influenza A in adults (Review)

Copyright $\ensuremath{\mathbb S}$ 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.



In the fourth updated review, we searched the Cochrane Central Register of Controlled Trials (CENTRAL) (The Cochrane Library 2008, issue 1), MEDLINE (2005 to April Week 4, 2008) and EMBASE (July 2005 to April 2008). There were no language or publication restrictions.

We ran the following search strategy on MEDLINE (a similar search strategy was used to search for trials on CENTRAL and EMBASE).

MEDLINE

1 exp INFLUENZA 2 influenza\$ 3 or/1-2 4 exp AMANTADINE 5 amantadine 6 exp RIMANTADINE 7 rimantadine 8 or/4-7 93 and 8 10 RANDOMIZED CONTROLLED TRIAL.pt. 11 CONTROLLED CLINICAL TRIAL.pt. 12 RANDOMIZED CONTROLLED TRIALS.sh. 13 RANDOM ALLOCATION.sh. 14 DOUBLE BLIND METHOD.sh. 15 SINGLE-BLIND METHOD.sh. 16 or/10-15 17 (ANIMAL not HUMAN).sh. 18 16 not 17 19 CLINICAL TRIAL.pt. 20 exp Clinical Trials 21 (clin\$ adj25 trial\$).ti,ab. 22 ((singl\$ or doubl\$ or trebl\$ or tripl\$) adj25 (blind\$ or mask \$)).ti,ab. 23 PLACEBOS.sh. 24 placebo\$.ti,ab. 25 random\$.ti,ab. 26 or/19-25 27 26 not 17 28 18 or 27 29 9 and 28

Data collection and analysis

Selection of studies

Two review authors (VD, TOJ) independently read all trials retrieved in the search and applied the inclusion criteria. VD and TOJ assessed trials fulfilling the review inclusion criteria for quality, and analysed results.

Data extraction and management

Two review authors (TOJ and DR) extracted data from included studies on standard forms. The procedure was supervised and arbitrated by VD. The following data were extracted, checked and recorded:

- 1. Characteristics of participants.
- 2. Number of participants.
- 3. Age, gender, ethnic group and risk category.

Characteristics of interventions

Type of antiviral, type of placebo, dose, treatment or prophylaxis schedule and length of follow up (in days).

Characteristics of outcome measures

- 1. Number and severity of influenza and ILI cases and deaths in amantadine/rimantadine and placebo groups.
- 2. Length of nasal shedding of influenza viruses or persistence in the upper airways.

Adverse effects

Four categories were used:

- 1. Gastrointestinal (GI) symptoms (nausea, vomiting, dyspepsia, diarrhoea and constipation).
- 2. Increased central nervous system (CNS) activity (insomnia, restlessness, light-headedness, nervousness and concentration problems).
- 3. Decreased CNS activity (malaise, depression, fatigue, vertigo and feeling drunk).
- 4. Dermatological changes (urticaria and rash).

(Adverse effect data were collected as the number of participants experiencing each (or any) adverse effect).

- 1. Number of withdrawals due to adverse effects.
- 2 Date of trial
- 3. Location of trial.
- 4. Sponsor of trial (specified, known or unknown).
- 5. Publication status.

Assessment of risk of bias in included studies

We carried out assessment of methodological quality for RCTs using criteria from the Cochrane Handbook for Systematic Reviews of Interventions (Higgins 2008). We assessed studies according to randomisation, generation of the allocation sequence, allocation concealment, blinding and follow up. We entered extracted data into Cochrane Review Manager software (RevMan 2008). Aggregation of data was dependent on the sensitivity and consistency of definitions of exposure, populations and outcomes used.

Assessment of trial quality was made according to the following criteria:

- 1. Generation of allocation schedule (defined as the methods of generation of the sequence which ensures random allocation).
- 2. Measure(s) taken to conceal treatment allocation (defined as methods to prevent selection bias, that is to say, to ensure that all participants have the same chance of being assigned to one of the arms of the trial. It protects the allocation sequence before and during allocation).
- 3. Number of drop-outs of allocated healthcare worker participants from the analysis of the trial (defined as the exclusion of any participants for whatever reason - deviation from protocol, loss to follow up, withdrawal, discovery of ineligibility, while the unbiased approach analyses all randomised participants in the originally assigned groups,

Amantadine and rimantadine for influenza A in adults (Review)

regardless of compliance with protocol - known as intention-to-treat analysis).

4. Measures taken to implement double-blinding (a single blind study is one in which observer(s) or subjects are kept ignorant of the group to which the subjects are assigned). When both the observer and the participants are kept ignorant of assignment the trial is called double-blind. Unlike allocation concealment, double blinding seeks to prevent ascertainment bias and protects the sequence after allocation).

For criteria 2, 3 and 4 there is empirical evidence that low quality in their implementation is associated with exaggerated trial results (Schulz 1995) and it is reasonable to infer a quality link between all four items.

The four criteria were assessed by answering the following questions:

Generation of allocation schedule

Did the review author(s) use?

- 1. Random number tables.
- 2. Computer random-number generator.
- 3. Coin tossing.
- 4. Shuffling of allocation cards.
- 5. Any other method which appeared random.

Concealment of treatment allocation

Which of the following was carried out?

- 1. There was some form of centralised randomisation scheme where details of an enrolled participant were passed to a trial office or a pharmacy to receive the treatment group allocation.
- 2. Treatment allocation was assigned by means of an on-site computer using a locked file which could be accessed only after inputting the details of the participant.
- 3. There were numbered or coded identical looking compounds which were administered sequentially to enrolled participants.
- 4. There were opaque envelopes, which had been sealed and serially numbered, utilised to assign participants to intervention(s).
- 5. A mixture of the above approaches including innovative schemes, provided the method appears impervious to allocation bias.
- 6. Allocation by alternation or date of birth or case record or day of the week or presenting order or enrolment order.

Concealment methods were described as 'adequate' for (1), (2), (3), (4) or (5). Method (6) was regarded as 'inadequate', as were trials using a system of random numbers or assignments. For some trials allocation was regarded as 'unclear' if only terms such as 'lists' or 'tables' or 'sealed envelopes' or 'randomly assigned' were mentioned in the text.

Exclusion of allocated participants from the analysis of the trial

- 1. Did the report mention explicitly the exclusion of allocated participants from the analysis of trial results?
- 2. If so did the report mention the reason(s) for exclusion? (if yes, specify).

Measures to implement double blinding

- 1. Did the report mention explicitly measures to implement and protect double blinding?
- 2. Did the author(s) report on the physical aspect of amantadine/ rimantadine administration, that is, appearances, colour, route of administration.

Arbitration procedure

There was no disagreement between TOJ and VD on the quality of trials, but DR was appointed as arbitrator.

Measures of treatment effect

The risk ratios (RR) of events comparing prophylaxis and placebo groups from the individual trials were combined using the DerSimonian and Laird (DerSimonian 1986) random-effects model to include between-trial variability. We carried out a sensitivity analysis of methods used, comparing our results obtained using the fixed-effect and random-effects models. In the prophylaxis trials efficacy was derived as 1-RR x 100 or the RR when not significant. Odds ratios (OR) were used to estimate association of adverse effects with exposure to antivirals. In treatment trials the choice of methods for combining the estimates of severity of influenza depended on the format in which the data were presented. We made comparisons between the mean duration of symptoms in the two groups, and methods for combining differences in means were used. Specifically, where the data were presented as the number of subjects with duration of symptoms beyond a cut-off time period these were presented as 'cases with fever at 48 hours'. The bewildering array of outcomes used in the treatment trials prevented us from using more than the 'cases with fever' outcome.

RESULTS

Description of studies

We identified 198 reports possibly fulfilling our inclusion criteria and retrieved 55 reports. We excluded 22 and classified one as pending translation from Polish. For descriptions see the 'Characteristics of included studies' table.

Prophylaxis trials

We identified 20 reports of 21 prophylaxis and safety trials fulfilling our inclusion criteria. We were unable to identify any unpublished trials, despite receiving nine letters and three electronic communications from manufacturers, authors and researchers.

The mean amantadine arm size was 327 individuals (median 140.5, 25th percentile 92.5, 75th percentile 348.2), the mean rimantadine arm size was 87 individuals (median 102, 25th percentile 63, 75th percentile 114) and the mean placebo arm size was 265 individuals (median 139, 25th percentile 99, 75th percentile 269). Differences in mean and median size were due to few bigger trials (Peckinpaugh 1970a; Peckinpaugh 1970b; Smorodintsev 1970) and several smaller ones.

The mean sample was 599 individuals (median 297, 25th percentile 202, 75th percentile 536). Mean length of follow up was 30 days (median 30 days, 25th percentile 16.5 days, 75th percentile 42 days). The duration of the epidemic was specified by only one trial (Kantor 1980) and was 49 days.

Amantadine and rimantadine for influenza A in adults (Review)

Copyright @ 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Cochrane Database of Systematic Reviews

The identified trials are listed below (using the name of the first author):

Brady 1990 Callmander 1968 Dolin 1982 Hayden 1981 Kantor 1980 Máté 1970 Millet 1982 Monto 1979 Muldoon 1976 Nafta 1970 Oker-Blom 1970 Payler 1984 Peckinpaugh 1970a Peckinpaugh 1970b Pettersson 1980 Plesnik 1977 Quarles 1981 Reuman 1989 Schapira 1971 Smorodintsev 1970 Wendel 1966

Treatment trials

We identified 13 published treatment trials (one by Máté 1970 contained both treatment and prophylaxis data). We were unable to identify any unpublished trials. The mean amantadine arm size was 80 individuals (median 63, 25th percentile 18.5, 75th percentile 90.2), the mean rimantadine arm size was 47 individuals (median 20, 25th percentile 11.5, 75th percentile 82.5) and the mean control arm size was 66 individuals (median 35.5, 25th percentile 13.5, 75th percentile 87.6). Again, differences in mean and median size were due to one bigger trial (Kitamoto 1968) and the others being smaller ones. The mean sample was 140 individuals (median 90.5, 25th percentile 29.7, 75th percentile 87.6). Mean length of follow up was 23 days (median 21 days, 25th percentile 10 days, 75th percentile 30 days).

Identified trials are listed below using the name of the first author and year of publication in the case of there being more than one trial by the same author. One trial (Hornick 1969) was broken down further into four sub-trials (see below for explanation).

Galbraith 1971 Hayden 1980 Hayden 1986 Hornick 1969a Hornick 1969b Hornick 1969c Hornick 1969d Ito 2000 Kitamoto 1968 Kitamoto 1971 Knight 1970 Máté 1970 Rabinovich 1969 Younkin 1983 van Voris 1981 Wingfield 1969

We identified 10 reports related to 11 trials which had been carried out during the 1968 to 1969 pandemic (Galbraith 1971; Kitamoto 1968; Knight 1970; Máté 1970; Muldoon 1976; Nafta 1970; Oker-Blom 1970; Peckinpaugh 1970a; Peckinpaugh 1970b; Schapira 1971; Smorodintsev 1970).

Risk of bias in included studies

Two review authors (VD, TOJ) assessed allocation method, allocation concealment, blinding and completeness of follow up. There were 30 trials in all, 28 of which considered either amantadine and/or rimantadine efficacy and two (Hayden 1981; Millet 1982) which considered adverse effects only. Twelve prophylaxis trials and seven treatment trials reported sufficient data on adverse effects. The quality of the prophylaxis trials was relatively good, considering their age. Among the 20 prophylaxis trials, 17 stated that the allocation method was randomised, although only four mentioned a particular method (Brady 1990; Monto 1979; Pettersson 1980; Reuman 1989) and two did not mention random allocation at all (Plesnik 1977; Schapira 1971). These two trials have therefore been classified as controlled clinical trials (CCTs) rather than RCTs. All prophylaxis trials were stated to be double-blind, with the exception of Payler 1984 which was open and had no placebo group (the comparison group was no intervention other than influenza vaccine at the beginning of the season). Among the 13 treatment trials, 11 stated that the allocation method was randomised and no trials mentioned a particular method. For Hornick's trials (Hornick 1969a; Hornick 1969b; Hornick 1969c; Hornick 1969d) there was no mention of random allocation at all. Very limited information was available for one trial (Ito 2000).

Major flaws in the reporting of trials were:

- 1. Lack of information on the completeness of follow up. In many trials there was a large difference between the number randomised and the number who actually participated.
- 2. Lack of detailed description of methods to conceal allocation with many trials just describing a "double-blind" procedure.
- 3. Frequent inconsistencies in the reporting of numerators and denominators in various arms of trials.
- 4. In the treatment trials, the use of a bewildering variety of outcomes, such as severity scores, of which none were alike.

A full description of all trials is available in the 'Characteristics of included studies' table.

Effects of interventions

We carried out nine comparisons:

Comparison A - oral amantadine compared to placebo in the prophylaxis of influenza or ILI.

Comparison B - oral rimantadine compared to placebo in the prophylaxis of influenza or ILI.

Comparison C - oral amantadine compared to oral rimantadine in the prophylaxis of influenza or ILI.

Comparison D - oral amantadine compared to placebo in the treatment of influenza.

Comparison E - oral rimantadine compared to placebo in the treatment of influenza.

Comparison F - oral amantadine compared to oral rimantadine in the treatment of influenza.

Amantadine and rimantadine for influenza A in adults (Review)

Copyright $\ensuremath{\mathbb S}$ 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.



Comparison G - oral or inhaled amantadine versus placebo or aspirin in the nasal viral shedding or persistence in upper airways at two to five days.

Comparison H - oral amantadine compared to control medication in the treatment of influenza.

Comparison I - inhaled amantadine compared to placebo in the treatment of influenza.

For comparisons A, B and C we analysed the effects on 'cases', stratified either as influenza (a defined set of signs and symptoms backed up by serological confirmation and/or isolation of influenza virus from nasal fluids) or clinical criteria alone (ILI) or asymptomatic cases (serological confirmation and/or isolation of influenza virus from nasal fluids without symptoms). The effects on nasal viral shedding were assessed by single studies: Reuman 1989(amantadine) and Dolin 1982 (rimantadine). We stratified comparisons on the basis of whether participants had received vaccination or not.

Additionally we assessed adverse effects in both comparisons.

In Comparisons A and B significant heterogeneity between the trial results was evident for both types of influenza analyses, so all results quoted are average treatment effects based on random-effects models. In Comparison A, amantadine prevented 61% (95% CI 35% to 76%) of influenza cases and 25% (95% CI 13% to 36%) of ILI cases. Both of these results are highly statistically significant (P < 0.001). There was no effect on asymptomatic cases (risk ratio (RR) 0.85; 95% CI 0.40 to 1.80), nor any difference in efficacy between unvaccinated and vaccinated individuals (RR 0.45; 95% CI 0.28 to 0.74 and RR 0.10; 95% CI 0.03 to 0.34). The effectiveness in unvaccinated subjects is significantly higher than that of placebo in unvaccinated subjects (RR 0.42; 95% CI 0.07 to 2.52), but not in vaccinated individuals 0.75 (95% CI 0.62 to 0.90).

In Comparison B rimantadine was not effective against either influenza (RR 0.28; 95% CI 0.08 to 1.08) or ILI (RR 0.65; 95% CI 0.35 to 1.20), however, analysis using a fixed-effect model shows significant protection against influenza and ILI in unvaccinated participants. Whilst these results are conventionally not statistically significant (P = 0.07 and P = 0.17, respectively), the estimates are based on only 688 individuals, and are of a very similar magnitude to those for amantadine. There was no effect on asymptomatic cases (RR 1.39; 95% CI 0.45 to 4.27), although this observation is based on one study only (Dolin 1982).

In Comparison C there is no evidence of a difference in efficacy between amantadine and rimantadine, although the confidence interval is quite wide (RR amantadine versus rimantadine 0.88. 95% CI 0.57 to 1.35). In comparisons B and C there were insufficient data to stratify by vaccine status of participants.

The 'all adverse effects' category includes all types and was derived from those trials which either did not report sufficient information to allow a more detailed classification or which presented aggregate data. Adverse effects incidence is reported in our meta-analysis as the number of participants with at least one event, thus the incidence of individual adverse effects cannot be summed to give the total with any adverse effect as more than one adverse event is likely to have taken place in the same individual during the trial. In Comparison A gastrointestinal symptoms (mainly nausea, odds ratio (OR) 2.56; 95% CI 1.37 to 4.79), insomnia and hallucinations (OR 2.54; 95% CI 1.50 to 4.31) and withdrawals from the trials because of adverse events (2.54; 95% CI 1.60 to 4.06) were significantly more common in participants who received amantadine than placebo. Analysis using a fixed-effect model shows a significant association with depression, insomnia and the 'all adverse events' category.

In Comparison B, rimantadine recipients were also more likely to experience 'all adverse effects' than placebo recipients (OR 1.96; 95% CI 1.19 to 3.22). However, there was no evidence of an increase in CNS-related effects with rimantadine, and withdrawal rates were similar in both groups.

The direct comparison of amantadine with rimantadine (Comparison C) confirmed that CNS adverse effects and withdrawal from trials were significantly more frequent among amantadine recipients than rimantadine recipients (CNS effects, OR 3.11; 95% CI 1.67 to 5.78; withdrawals OR 2.49; 95% CI 1.26 to 4.93).

Thus rimantadine may be no less efficacious but safer than amantadine in preventing cases of influenza in healthy adults. Readers should bear in mind that the study sizes of the safety trials of rimantadine are considerably smaller than those of amantadine, so that the conclusions that can be drawn for rimantadine are somewhat less certain than those for amantadine.

We considered meta-analysing symptoms outcome data to further inform the assessment of the effects of amantadine and rimantadine in the treatment role. When we tabulated the outcome typology we discovered that such a meta-analysis would be impossible as can be seen from Table 1.

We resorted to using duration of fever (defined as a temperature greater than 37 °C) as the only common outcome. One obvious cost of this approach is the possible confounding effect of the presence of fever for a variable length of time prior to and after entry to the study (and hence at the moment of commencement of treatment). However, if random allocation had been properly carried out, this effect should disappear.

In Comparison D amantadine significantly shortened duration of fever compared to placebo (by 0.99 days; 95% CI 0.71 to 1.26). The meta-analysis is based on 542 subjects (250 in the amantadine and 292 in the placebo arm). Where time to fever clearance data were not available (as in van Voris 1981 and Wingfield 1969), a dichotomous outcome was used (cases with fever at 48 hours). This comparison showed that amantadine was significantly better than placebo (RR 0.21; 95% CI 0.07 to 0.66). However, there was no effect on nasal shedding or persistence of influenza A viruses in the upper airways after up to five days of treatment (RR 0.97; 95% CI: 0.76 to 1.24).

In Comparison E rimantadine shortened duration of fever compared to placebo (by 1.24 days; 95% CI -0.76 to -1.71). There were a significantly higher number of afebrile cases 48 hours after commencing rimantadine treatment (RR 0.16; 95% CI 0.05 to 0.53). However, there was no effect on nasal shedding or persistence of influenza A viruses in the upper airways after up to five days of treatment (RR 0.68; 95% CI 0.30 to 1.53), although this finding may be due to the small number of observations in this comparison (152) and is sensitive to analysis using a fixed-effect model.



The few data available directly comparing amantadine and rimantadine for treatment (Comparison F) showed that the efficacy of the two drugs was comparable, although the confidence intervals are very wide (for example, cases with fever at 48 hours RR 0.99; 95% CI 0.23 to 4.37).

In contrast to the increased adverse effect rates for prophylaxis, there was no evidence that amantadine recipients had higher adverse effect rates than placebo recipients (Comparison D), but data were only available from three trials (Kitamoto 1968; Kitamoto 1971; van Voris 1981 with combined denominator of 491) and the association with decreased CNS activity is sensitive to the application of a fixed-effect model. There were very few data available for the assessment of adverse effects of rimantadine for treatment (45 participants in Hayden 1986 and van Voris 1981) or the direct comparison between amantadine and rimantadine (33 participants in van Voris 1981).

In comparison G the effects of oral or inhaled amantadine on shedding of influenza A viruses are still not significant (RR 0.94; 95% CI 0.74 to 1.19), despite meta-analysis of five studies with a combined denominator of 237 observations.

Readers of this review should bear in mind that the difference in incidence of adverse effects is of importance, rather than the estimated incidence itself, as the adverse effects reported with these drugs are very similar to the clinical manifestations of influenza infection.

Overall both drugs appear to be effective and well-tolerated, but the evaluation of the effects of rimantadine was carried out on a very small population.

Insufficient data were available to analyse the relationships of dose (or duration) of treatment and clinical or virological effects. However, other data suggest that equivalent doses of amantadine and rimantadine at steady-state are associated with similar plasma concentrations and similar total clearance values (Aoki 1998).

We carried out further comparisons (H and I).

In Comparison H, based on Younkin 1983 and Ito 2000, standard medications (aspirin and other antipyretic or anti-inflammatory drugs or antibiotics were equally effective compared with amantadine in reducing the length of fever (mean difference (MD) random-effects model 0.25; 95% CI - 0.37 to 0.87). This observation is based on 78 individuals and in the trial by Ito 2000 amantadine was given at the lower dose of 100 mg. Aspirin and the other antipyretic drugs appear to be as potent as amantadine in treating symptoms, however they do not inhibit viral replication and as such remain a symptomatic remedy.

In comparison I (based on the Hayden 1980 trial) inhaled amantadine was no more efficacious than placebo in bringing down the respiratory or constitutional symptom score (MD -1.00; 95% CI -3.64 to 1.64 and -2.00; 95% CI -16.98 to 12.98, respectively). This comparison is based on small numbers of participants (20). Not surprisingly, amantadine caused significantly more nasal irritation (RR 12.50; 95% CI 1.09 to 143.43). Inhaled amantadine does not appear to be particularly effective but has a high incidence of local adverse effects, which would make compliance difficult.

Neither comparison showed an effect on nasal shedding or persistence of influenza A viruses in the upper airways after up to

five days of treatment, although the interpretation of Comparisons H and I is made difficult by the small numbers involved and the absence of multiple trials.

All trials tested the effects of amantadine and rimantadine on a wide variety of influenza A viruses. None tested the effects on influenza B, on which the molecules are known to be ineffective. No trial tested the role of the compounds on workplace outbreak control, which is a pity considering the trial settings (prisons, factories, schools, barracks).

Some trials are likely to have included individuals who took aspirin to relieve symptoms (especially in the treatment trials). However the effects of this potential confounder should have been eliminated by the process of randomisation.

All trials commenced administration of the compounds within a reasonable time frame. Treatment started at the latest 48 hours after positive identification of the first case in the population and prophylaxis when the results of surveillance made it reasonable to do so.

No trials assessed onset of resistance, but data in one study demonstrated that 10% to 27% of patients treated with amantadine secreted drug-resistant virus within four to five days of commencing treatment (Aoki 1998).

Separate analysis of the 11 pandemic trials did not affect our findings. Finally, we considered carrying out sub-analysis by dose (100, 200, 300 mg daily), but decided against this given the small size of the resulting meta-analysis. We will re-consider this policy if any further data become available.

DISCUSSION

The results of our review show that both amantadine and rimantadine are efficacious and relatively safe in the prophylaxis and treatment of influenza A symptoms. The role of amantadine in prophylaxis of symptoms (61% effective) and treatment (shortens duration of illness by one day) is beyond question and does not need to be investigated further compared to placebo. Rimantadine appears equally efficacious in prophylaxis (72%), but in direct comparison with placebo, when a random-effects model is applied, the lower bound of the 95% CI does not achieve statistical significance.

There are two explanations for this difference in the significance of the findings. The first is that the rimantadine result is a 'false negative'. This idea is supported by noting that its average efficacy is both large and similar to amantadine, and that there have been many fewer participants in rimantadine trials than amantadine trials (there are clinical data for approximately only 700 rimantadine compared to 2500 amantadine recipients in the review).

The second explanation is centred on trial heterogeneity. If a fixed-effect analysis is used (effectively ignoring the heterogeneity) then the difference between rimantadine and placebo for the prophylaxis of influenza cases is highly significant (P value less than 0.001 for both outcomes). All of the analyses of influenza outcomes demonstrated excessive variation in the results of the trials. Such a pattern has been noted in other reviews of preventive procedures, such as influenza and cholera vaccination, and may reflect differences between the trial populations to natural



exposure and immunity to influenza A and other similar viruses. We have not been able to explain this heterogeneity in this systematic review.

Rimantadine was also seen to be equally therapeutically efficacious, shortening duration of fever by just over one day. However, again, this observation is based on 82 subjects only.

There is a marked difference between the two drugs in the capacity to prevent influenza and the capacity to prevent influenza-likeillness (ILI). The practical importance of this difference, which is rarely explained to the public, is that neither can be used to good effect against ILI, which is the clinical picture presenting to both the patient and the doctor. In the absence of a likely influenza diagnosis their routine use is not to be recommended. This conclusion is also supported by widespread evidence of resistance to both amantadine and rimantadine (Bright 2005).

There do not appear to be significant differences in effectiveness in either role between the two compounds, although again our comparisons are based on small numbers with large confidence intervals.

Our conclusions must be tempered by our finding of a lack of effect of both compounds both on influenza A cases with no clinical symptoms (asymptomatic) and on viral excretion (clearance) from the upper airways (although viral concentration in nasal mucus may be reduced). Both compounds are effective in preventing or treating symptoms, but have lower effectiveness in preventing infection and probably transmission, an observation made in one of the trials included in the review (Monto 1979). As a consequence, it is likely that the estimates of clinical efficacy and effectiveness presented in this review are optimistic. This finding is of crucial importance in planning the use of both compounds in a situation of very high viral circulation and infectivity (such as a serious epidemic or pandemic). In addition, symptom relief may lead to convalescing subjects who are still infected and infectious increasing viral transmission in the community. On the basis of this evidence the WHO recommendations should be redrafted to include the use of amantadine and rimantadine only in emergency situations when all other measures fail.

The safety profile of the two drugs appears significantly different in prophylaxis, with rimantadine causing significantly fewer central nervous system (CNS) adverse effects than amantadine and fewer withdrawals from the trials. Although these observations are based on smaller numbers of rimantadine recipients, amantadine definitely causes signs of significantly increased CNS activity, an effect which is not easily acceptable by healthy adults, especially in employment which requires concentration and mental fitness. Rimantadine has a different pharmacokinetic profile from amantadine, reaching prophylactic concentration in the nasal mucus at much lower plasma concentrations than amantadine. There was a tendency for lower doses of amantadine (100 mg daily) to cause fewer adverse effects than higher doses at the cost of lower effectiveness (data not shown).

We conclude that from the available evidence, rimantadine appears the better choice for individual protection in emergencies.

In future, more attention should be paid to the assessment of adverse events of the two compounds, particularly those of rimantadine which at present are based on relatively small numbers.

The quality of the trials was not good with significant numbers of studies failing to give adequate descriptions of methods and of results. This may be in part due to the number of older trials in the review. Both quality of trial conducting and reporting should be improved and adverse effects and case outcome definitions should be standardised. Finally, the bewildering array of outcome definitions used in treatment studies made the task of meta-analysis difficult and led to a great loss of information.

AUTHORS' CONCLUSIONS

Implications for practice

Both drugs should be used only in emergency situations when all other measures fail.

Implications for research

Given our findings we do not believe any further research should be carried out on these compounds.

ACKNOWLEDGEMENTS

The review authors would like to thank Drs Aoki, Couch, Hayden and Monto for helpful comments; Dr van der Wouden for providing the Máté 1970 trial report; and Mrs Carol Hobbs, Ms Ruth Foxlee, Ann-Charlott Johansson and Jessika Wejfalk for assistance with trial identification and retrieval. The review authors wish to acknowledge Jon Deeks for his contribution as a co-author on the original and subsequent updates until 2004. The review authors also wish to thank Melanie Rudin for her assistance with the administrative organising of the 2005 update and for keeping track of the information, and to thank the following people for commenting on earler versions of the review: Amy Zelmer, Douglas Fleming, Kuldeep Kumar and Allen Cheng. Finally, the review authors wish to thank the following people for commenting on the 2008 updated draft: Mina Mahdian, Fred Aoki, Mark Jones, and Allen Cheng.

REFERENCES

References to studies included in this review

Brady 1990 {published data only}

Brady MT, Sears SD, Pacini DL, Samorodin R, DePamphilis J, Oakes M, et al. Safety and prophylactic efficacy of lowdose rimantadine in adults during an influenza A epidemic Antimicrobial. *Agents and Chemotherapy* 1990;**34**:1633-66.

Callmander 1968 {published data only}

Callmander E, Hellgren L. Amantadine hydrochloride as a prophylactic in respiratory infections. A double-blind investigation of its clinical use and serology. *Journal of Clinical Pharmacology* 1968;**8**:186-9.

Dolin 1982 {published data only}

Dolin R, Reichman RC, Madore HP, Maynard R, Linton PN, Webber-Jones J. A controlled trial of amantadine and rimantadine in the prophylaxis of influenza A infection. *New England Journal of Medicine* 1982;**307**:580-4.

Galbraith 1971 {published data only}

Galbraith AW, Oxford JS, Schild GC, Potter CW, Watson GI. Therapeutic effect of 1-adamantanamine hydrochloride in naturally occurring influenza A 2/Hong Kong infection - a controlled double-blind study. *Lancet* 1971;**2**:113-5.

Hayden 1980 {published data only}

Hayden FG, Hall WJ, Douglas RG. Therapeutic effects of aerosolized amantadine in naturally acquired infection due to influenza A virus. *Journal of Infectious Diseases* 1980;**141**:535-42.

Hayden 1981 {published data only}

Hayden FG, Gwaltney JM, van de Castle RL, Adams KF, Giordani B. Comparative toxicity of amantadine hydrochloride and rimantadine hydrochloride in healthy adults. *Antimicrobial Agents and Chemotherapy* 1981;**19**:226-33.

Hayden 1986 {published data only}

Hayden FG, Monto AS. Oral rimantadine hydrochloride therapy of influenza A virus H3 N2 subtype infection in adults. *Antimicrobial Agents and Chemotherapy* 1986;**29**:339-41.

Hornick 1969a {published data only}

Hornick RB, Togo Y, Mahler S, Iezzoni D. Evaluation of amantadine hydrochloride in the treatment of A2 influenzal disease. *Bulletin of the World Health Organization* 1969;**41**:671-6.

Hornick 1969b {published data only}

Hornick RB, Togo Y, Mahler S, Iezzoni D. Evaluation of amantadine hydrochloride in the treatment of A2 influenzal disease. *Bulletin of the World Health Organization* 1969;**41**:671-6.

Hornick 1969c {published data only}

Hornick RB, Togo Y, Mahler S, Iezzoni D. Evaluation of amantadine hydrochloride in the treatment of A2 influenzal disease. *Bulletin of the World Health Organization* 1969;**41**:671-6.

Hornick 1969d {published data only}

Hornick RB, Togo Y, Mahler S, Iezzoni D. Evaluation of amantadine hydrochloride in the treatment of A2 influenzal disease. *Bulletin of the World Health Organization* 1969;**41**(Trial no AMANT18):671-6.

Ito 2000 {published data only}

Ito S, Iijima N, Kanemaki K, Hayashi S, Hujii S, Watanabe T, et al. Therapeutic efficacy of amantadine hydrochloride in patients with epidemic influenza A virus infection. *Nihon Kokyuki Gakkai zasshi (The Journal of the Japanese Respiratory Society)* 2000;**38**(12):897-902.

Kantor 1980 {published data only}

Kantor RJ, Potts DW, Stevens D, Noble GR. Prevention of influenza A/USSR/77 (H1N1): an evaluation of the side effects and efficacy of amantadine in recruits at Fort Sam Houston. *Military Medicine* 1980;**145**:312-5.

Kitamoto 1968 {published data only}

Kitamoto O. Therapeutic effectiveness of amantadine hydrochloride in influenza A2 - double blind studies. *Japanese Journal of Tuberculosis and Chest Diseases* 1968;**15**:17-26.

Kitamoto 1971 {published data only}

Kitamoto O. Therapeutic effectiveness of amantadine hydrochloride in naturally occurring Hong Kong influenza double-blind studies. *Japanese Journal of Tuberculosis and Chest Diseases* 1971;**17**:1-7.

Knight 1970 {published data only}

Knight V, Fedson D, Baldini J, Douglas RG, Couch RB. Amantadine therapy of epidemic influenza A2 (Hong Kong). *Infection and Immunity* 1970;**1**:200-4.

Máté 1970 {published data only}

Máté J, Simon M, Juvancz I, Takátsy G, Hollós I, Farkas E. Prophylactic use of amantadine during Hong Kong influenza epidemic. *Acta Microbiologica Academiae Scientiarum Hungaricae* 1970;**17**:285-96.

Millet 1982 {published data only}

Millet VM, Dreisbach M, Bryson YJ. Double-blind controlled study of central nervous system side effects of amantadine, rimantadine and chlorpheniramine. *Antimicrobial Agents and Chemotherapy* 1982;**21**(1):1-4.

Monto 1979 {published data only}

Monto AS, Gunn RA, Bandyk MG, King CL. Prevention of Russian influenza by amantadine. *Journal of the American Medical Association* 1979;**241**:1003-7.

Muldoon 1976 {published data only}

Muldoon RL, Stanley ED, Jackson GG. Use and withdrawal of amantadine chemoprophylaxis during epidemic influenza A. *American Review of Respiratory Disease* 1976;**113**:487-91.



Nafta 1970 {published data only}

Nafta I, Turcanu AG, Braun I, Companetz W, Simionescu A, Birţ E, et al. Administration of amantadine for the prevention of Hong Kong influenza. *Bulletin of the World Health Organization* 1970;**42**(3):423-7.

Oker-Blom 1970 {published data only}

Oker-Blom N, Hovi T, Leinikki P, Palosuo T, Pettersson R, Suni J. Protection of man from natural infection with influenza A2 Hong Kong virus by amantadine: a controlled field trial. *British Medical Journal* 1970;**3**:676-8.

Payler 1984 {published data only}

Payler DK, Purdham PA. Influenza A prophylaxis with amantadine in a boarding school. *Lancet* 1984;**1**:502-4.

Peckinpaugh 1970a {published data only}

Peckinpaugh RO, Askin FB, Pierce WE, Edwards EA, Johnson DP, Jackson GG. Field studies with amantadine: acceptability and protection. AMANTADINE 42. *Annals of the New York Academy of Sciences* 1970;**173**:62-73.

Peckinpaugh 1970b {published data only}

Peckinpaugh RO, Askin FB, Pierce WE, Edwards EA, Johnson DP, Jackson GG. Field studies with amantadine: acceptability and protection. AMANTADINE 42. *Annals of the New York Academy of Sciences* 1970;**173**:62-73.

Pettersson 1980 {published data only}

Pettersson RF, Hellstrom PE, Penttinen K, Pyhala R, Tokola O, Vartio T, et al. Evaluation of amantadine in the prophylaxis of influenza A (H1N1) virus infection: a controlled field trial among young adults and high-risk patients. *Journal of Infectious Diseases* 1980;**142**:377-83.

Plesnik 1977 {published data only}

Plesnik V, Heinz F, Bindas B, Cechová D, Eliásová J, Galetková A, et al. Controlled study of influenza prophylaxis using amantadine VUFB. *Chzekolsovak Epidemiologie Mikrobiologie, Immunologie* 1977;**26**(4):216-26.

Quarles 1981 {published data only}

Quarles JM, Couch RB, Cate TR, Goswick CB. Comparison of amantadine and rimantadine for prevention of type A (Russian) influenza. *Antiviral Research* 1981;**1**:149-55.

Rabinovich 1969 {published data only}

Rabinovich S, Baldini JT, Bannister R. Treatment of influenza. The therapeutic efficacy of rimantadine HC1 in a naturally occurring influenza A2 outbreak. *The American Journal of the Medical Sciences* 1969;**257**(5):328-35.

Reuman 1989 {published data only}

Reuman PD, Bernstein DI, Keefer MC, Young EC, Sherwood JR, Schiff GM. Efficacy and safety of low dosage amantadine hydrochloride as prophylaxis for influenza A. *Antiviral Research* 1989;**11**:27-40.

Schapira 1971 {published data only}

Schapira M, Oxford JS, Galbraith AW. A study of 1adamantamine hydrochloride during the 1970 Hong Kong influenza epidemic. *Journal of the Royal College of General Practitioners* 1971;**21**:695-7.

Smorodintsev 1970 {published data only}

Smorodintsev AA, Karpuhin GI, Zlydnikov DM, Malyseva AM, Svecova EG, Burov SA, et al. The prophylactic effectiveness of amantadine hydrochloride in an epidemic of Hong Kong influenza in Leningrad in 1969. *Bulletin of the World Health Organization* 1970;**42**:865-72.

van Voris 1981 {published data only}

van Voris LP, Betts RF, Hayden FG, Christmas WA, Douglas RG. Successful treatment of naturally occurring influenza A/ USSR/77 H1N1. *Journal of the American Medical Association* 1981;**245**:1128-31.

Wendel 1966 {published data only}

Wendel HA, Snyder MT, Pell S. Trial of amantadine in epidemic influenza. *Clinical Pharmacology and Therapeutics* 1966;**7**:38-43.

Wingfield 1969 {published data only}

Wingfield WL, Pollack D, Grunert RR. Therapeutic efficacy of amantadine HC1 and rimantadine HC1 in naturally occurring influenza A2 respiratory illness in man. *New England Journal of Medicine* 1969;**281**:579-84.

Younkin 1983 {published data only}

Younkin SW, Betts RF, Roth FK, Douglas RG. Reduction in fever and symptoms in young adults with influenza A/Brazil/78 H1N1 infection after treatment with aspirin or amantadine. *Antimicrobial Agents and Chemotherapy* 1983;**23**:577-82.

References to studies excluded from this review

Aoki 1986 {published data only}

Aoki FY, Stiver HG, Sitar DS, Hammond GW, Milley EV, Vermeersch C, et al. Potential of influenza vaccine and amantadine to prevent influenza A illness in Canadian forces personnel 1980-1983. *Military Medicine* 1986;**151**:459-65.

Atmar 1990 {published data only}

Atmar RL, Greenberg SB, Quarles JM, Wilson SZ, Tyler B, Feldman S, et al. Safety and pharmacokinetics of rimantadine small-particle aerosol. *Antimicrobial Agents and Chemotherapy* 1990;**34**:2228-33.

Baker 1969 {published data only}

Baker LM, Shock MP, Iezzoni DG. The therapeutic efficacy of Symmetrel (amantadine hydrochloride) in naturally occurring influenza A2 respiratory illness. *Journal of the American Osteopathic Association* 1969;**68**:1244-50.

Bricaire 1990 {published data only}

Bricaire F, Hannoun C, Boissel JP. Prevention of influenza A. Effectiveness and tolerance of rimantadine hydrochloride [Prévention de la grippe A. Efficacité et tolérance du chlorhydrate de rimantadine]. *Presse Medicale* 1990;**19**(2):69-72.

Bryson 1980 {published data only}

Bryson YJ, Monahan C, Pollack M, Shields WD. A prospective double-blind study of side effects associated with the administration of amantadine for influenza A virus prophylaxis. *Journal of Infectious Diseases* 1980;**141**:543-7.

Clover 1986 {published data only}

Clover RD, Crawford SA, Abell TD, et al. Effectiveness of rimantadine prophylaxis of children within families. *American Journal of Diseases of Children* 1986;**140**:706-9.

Crawford 1988 {published data only}

Crawford SA, Clover RD, Abell TD, Ramsey CN, Glezen WP, Couch RB. Rimantadine prophylaxis in children: a follow-up study. *Pediatric Infectious Disease Journal* 1988;**7**:379-83.

Dawkins 1968 {published data only}

Dawkins AT Jr, Gallager LR, Togo Y, Hornick RB, Harris BA. Studies on induced influenza in man. II. Double-blind study designed to assess the prophylactic efficacy of an analogue of amantadine hydrochloride. *JAMA* 1968;**203**(13):1095-9.

Finklea 1967 {published data only}

Finklea JF, Hennessy AV, Davenport FM. A field trial of amantadine prophylaxis in naturally-occurring acute respiratory illness. *American Journal of Epidemiology* 1967;**85**(3):403-12.

Galbraith 1969 {published data only}

Galbraith AW, Oxford JS, Schild GC, Watson GI. Study of 1adamantanamine hydrochloride used prophylactically during the Hong Kong influenza epidemic in the family environment. *Bulletin of the World Health Organization* 1969;**41**:677-82.

Galbraith 1973 {published data only}

Galbraith AW, Schild GC, Potter CW, Watson GI. The therapeutic effect of amantadine in influenza occurring during the winter of 1971-2 assessed by double-blind study. *Journal of the Royal College of General Practitioners* 1973;**23**:34-7.

Hayden 1983 {published data only}

Hayden FG, Hoffman HE, Spyker DA. Differences in side effects of amantadine hydrochloride and rimantadine hydrochloride relate to differences in pharmacokinetics. *Antimicrobial Agents and Chemotherapy* 1983;**231**(3):458-64.

Hayden 1985 {published data only}

Hayden FG, Minocha A, Spyker DA, Hoffman HE. Comparative single-dose pharmacokinetics of amantadine hydrochloride and rimantadine hydrochloride in young and elderly adults. *Antimicrobial Agents and Chemotherapy* 1985;**28**(2):216-21.

Hayden 1989 {published data only}

Hayden FG, Belshe RB, Clover RD, Hay AJ, Oakes MG, Soo W. Emergence and apparent transmission of rimantadine-resistant influenza A virus in families. *New England Journal of Medicine* 1989;**321**:1696-702.

Hayden 1991 {published data only}

Hayden FG, Sperber SJ, Belshe RB, Clover RD, Hay AJ, Pyke S. Recovery of drug-resistant influenza A virus during therapeutic use of rimantadine. *Antimicrobial Agents and Chemotherapy* 1991;**35**:1741-7.

Knight 1969 {published data only}

Knight V, Fedson D, Baldini J, Douglas RG, Couch RB. Amantadine therapy of epidemic influenza A2-Hong Kong. *Antimicrobial Agents and Chemotherapy* 1969;**9**:370-1.

O'Donoghue 1971 {published data only}

O'Donoghue JM, Ray CG, Terry DW, Beaty HN. Prevention of nosocomial influenza infection with Amantadine. *American Journal of Epidemiology* 1972;**97**:276-82.

Rose 1980 {published data only}

Rose HJ. The use of amantadine and influenza vaccine in a type A influenza epidemic in a boarding school. *Journal of the Royal College of General Practitioners* 1980;**30**:619-21.

Smorodinstev 1970a {published data only}

Smorodintsev AA, Karpuhin GI, Zlydnikov DM, Malyseva AM, Svecova EG, Burov SA, et al. The prospect of amantadine for prevention of influenza A2/Hong Kong epidemics in January-February 1969 in Leningrad. *Annals of the New York Academy of Science* 1970;**173**:44-73.

Smorodintsev 1970b {published and unpublished data}

Smorodintsev AA, Zlydnikov DM, Kiseleva AM, Romanov JA, Kazantev AP, Rumovsky VI. Evaluation of amantadine in artificially induced A2 and B influenza. *JAMA* 1970;**213**:1448-54.

Togo 1968 {published data only}

Togo Y, Hornick RB, Dawkins AT Jr. Studies on induced influenza in man. I. Double-blind studies designed to assess prophylactic efficacy of amantadine hydrochloride against a2/Rockville/1/65 strain. *JAMA* 1968;**203**(13):1089-94.

Togo 1970 {published data only}

Togo Y, Hornick RB, Felitti VJ, Kaufman ML, Dawkins AT, Kilpe VE, et al. Evaluation of therapeutic efficacy of amantadine in patients with naturally occurring A2 influenza. *JAMA* 1970;**211**:1149-56.

References to studies awaiting assessment

Tkaczewski 1972 {published data only}

Tkaczewski W, Niedzielska H, Denys A, Bochenska J, Kulawczyk M, Szymanski L, et al. Clinical evaluation of remantadine in the treatment of influenza. *Polski Tygodnik Lekarski* 1972;**27**(42):1650-2.

Additional references

Aoki 1998

Aoki FY. Amantadine and rimantadine. In: Nicholson, Hay, Webster editor(s). Textbook of Influenza. London: Blackwell, 1998:457-76.



ARI Group 1998

Cochrane Acute Respiratory Infections Group. Search strategy for specialised register. The Cochrane Library. Oxford: Update Software, 1998; Vol. Issue 1.

Bright 2005

Bright RA, Medina MJ, Xu X, Perez-Oronoz G, Wallis TR, Davis XM, et al. Incidence of adamantane resistance among influenza A (H3N2) viruses isolated worldwide from 1994 to 2005: a cause for concern. *Lancet* 2005;**366**:1175-81.

DerSimonian 1986

DerSimonian R, Laird N. Meta-analysis in clinical trials. *Controlled Clinical Trials* 1986;**7**:177-88.

Higgins 2008

Higgins JPT, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions Version 5.0.0 [updated February 2008]. In: Higgins JPT, Green S editor(s). The Cochrane Collaboration. John Wiley and Sons, Ltd, 2008.

RevMan 2008 [Computer program]

The Nordic Cochrane Centre, The Cochrane Collaboration. Review Manager (RevMan). Version 5.0. Copenhagen: The Nordic Cochrane Centre, The Cochrane Collaboration, 2008.

Schulz 1995

Schulz KF, Chalmers I, Hayes RJ, Altman DG. Empirical evidence of bias. Dimensions of methodological quality associated with estimates of treatment effects in controlled trials. *JAMA* 1995;**273**:408-12.

CHARACTERISTICS OF STUDIES

Characteristics of included studies [ordered by study ID]

Brady 1990

WHO 2005

World Health Organisation. Global agenda on influenza surveillance and control. http://www.who.int/csr/disease/ influenza/csrinfluenzaglobalagenda/en/print.html (accessed 10 October 2005).

References to other published versions of this review

Jefferson 1999

Jefferson TO, Demicheli V, Deeks JJ, Rivetti D. Amantadine and rimantadine for preventing and treating influenza A in adults. *Cochrane Database of Systematic Reviews* 1999, Issue 2. [DOI: 10.1002/14651858.CD001169.pub3]

Jefferson 2002

Jefferson TO, Demicheli V, Deeks JJ, Rivetti D. Amantadine and rimantadine for preventing and treating influenza A in adults. *Cochrane Database of Systematic Reviews* 2002, Issue 3. [DOI: 10.1002/14651858.CD001169.pub3]

Jefferson 2004

Jefferson TO, Demicheli V, Deeks JJ, Rivetti D. Amantadine and rimantadine for preventing and treating influenza A in adults. *Cochrane Database of Systematic Reviews* 2004, Issue 3. [DOI: 10.1002/14651858.CD001169.pub3]

Jefferson 2006

Jefferson T, Demicheli V, Rivetti D, Jones M, Di Pietrantonj C, Rivetti A. Antivirals for influenza in healthy adults: systematic review. *The Lancet* 2006;**367**(9507):303-13. [DOI: 10.1016/ S0140-6736(06) 67970-1.]

| Methods | Prophylaxis, randomised, double-blind, controlled trial of rimantadine during an epidemic of influenz A/Leningrad/87 [H3N2] virus | | |
|---------------|---|--|--|
| Participants | 228 healthy, not previously vaccinated, adult volunteers aged 18 to 55 | | |
| Interventions | Participants were randomised to receive either rimantadine 100 mg daily or placebo for 6 weeks | | |
| Outcomes | Laboratory: paired sera were taken from all participants at the beginning and the end of the study. Within-trial surveillance was carried out on a weekly basis and cases were defined on the basis of sero- conversion and a pre-defined list of symptoms and signs. | | |
| | The study reports separately on the efficacy of asymptomatic cases of influenza (diagnosed from a rise in antibody titres). Viral isolation took place by nasal washout | | |
| Notes | Brady is a clearly written and well-reported trial (with the exception of the minor discrepancy between text and tables on the affiliation of drop-outs). | | |
| | Randomisation was computer-generated and allocation was concealed with a centralised scheme. Ad- ditionally, intention-to-treat analysis is clearly stated in the text | | |
| Risk of bias | | | |

Amantadine and rimantadine for influenza A in adults (Review)

Copyright $\ensuremath{\mathbb S}$ 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.



Brady 1990 (Continued)

| Bias | Authors' judgement | Support for judgement |
|---|--------------------|-----------------------|
| Allocation concealment (selection bias) | Low risk | A - Adequate |

Callmander 1968

| Piac | Authors' judgement Support for judgement | |
|---------------|---|--|
| Risk of bias | | |
| Notes | The practices of randomisation, allocation and concealment are not further defined, making it impossi- ble to assess methodological rigour although as the distribution of sex and age was checked and found to be similar, randomisation is likely to have been satisfactory | |
| Outcomes | Efficacy: ILI cases (from a symptoms list) in each arm and a symptom score (reported in Table 1 without an indication of time of intensity). Surveillance for adverse effects (systemic) was carried out. A list of symptoms (without a denominator) is reported in Table 2 | |
| Interventions | The intervention arm received 100 mg of amantadine hydrochloride twice daily and the control arm (a not further described placebo) | |
| Participants | The age range of the 94 volunteer participants is 20 to 60 years (44 male and 50 female) | |
| Methods | Randomised controlled trial conducted in a community, including some military personnel. During th period of the trial there was considerable influenza A2 (Leningrad) activity | |

| Blas | Authors' Judgement | Support for Judgement |
|--|--------------------|-----------------------|
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

| Dolin 1982 | | | |
|---------------|--|--|--|
| Methods | Prophylaxis, randomised, double-blind, placebo controlled trial carried out in Burlington Ver- mont, USA. The trial was commenced on 10 January 1981 during an outbreak of influenza A/ Bangkok/1/79H3N2 and A/Brazil/11/78H1N1 detected by surveillance (see Figure 1 in the text of the tri- al report) | | |
| Participants | Participants initially were 450 healthy non-vaccinated volunteers aged 18 to 45 (mean age 25.6 + 0.45 years). The final total of participants was 378 (132 in the placebo arm, 133 in the rimantadine arm and 113 in the amantadine arm) | | |
| Interventions | Amantadine 200 mg daily or rimantadine 200 mg or placebo | | |
| Outcomes | Efficacy: case definition was based on a list of symptoms plus virus isolation or a rise in serum antibody titres to influenza A. Table 1 presents both ILI cases and cases defined on the basis of laboratory confirmation. | | |
| | The study reports separately the efficacy on asymptomatic cases of influenza (diagnosed from a rise in antibody titres) | | |
| Notes | Although a well-written report, no real information is given on random allocation, blinding and con- cealment. Intention-to-treat analysis was not carried out | | |

Dolin 1982 (Continued)

Risk of bias

| Bias | Authors' judgement | Support for judgement |
|---|--------------------|-----------------------|
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

Galbraith 1971

| Methods | Treatment trial carried out in December 1969 to January 1970, at the time of an epidemic (possible pandemic) caused by a variant of A2/Hong Kong/68 | | |
|--|--|-----------------------|--|
| Participants | Participants were unvaccinated family members aged more than 2 years recruited by 57 family doctors in the United Kingdom | | |
| Interventions | 153 participants with laboratory-confirmed diagnosis of influenza A2 were randomised to receive either doses appropriate to their ages: for adults amantadine 200 mg a day (n = 72, mean age 37.4 years), or placebo (n = 81, mean age 39.1). Treatment was commenced within 48 hours of symptoms and continued for 7 days | | |
| Outcomes | Efficacy: outcomes are clinical (Tables 2 and 3) and serological (Table 4 and 5). In our meta-analysis, we have included the time of duration of fever (in days after commencement of treatment (Table 2) approximating the standard deviation of duration (not reported in the text) from the P value reported in the table. No adverse effect is mentioned or reported in the text | | |
| Notes | The authors conclude that amantadine treatment was effective in controlling fever, but no other symp- toms, possibly due to lack of sensitivity of surveillance methods. Although randomisation is clearly mentioned, no detailed description of allocation and concealment is given, making its assessment im- possible | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear | |

Hayden 1980

| Randomised, double-blind, placebo controlled treatment trial of inhaled (20 mg daily) amantadine |
|--|
| 20 participants |
| Participants were randomised to receive either amantadine (n = 9, mean 19.1 years) or diluted water placebo (n = 11, mean age 20.3 years) within 48 hours of developing symptoms for a duration of 4 days |
| Laboratory: influenza A/Texas/77[H3N2] and influenza A/USSR/77[H1N1] caused infection in the partic- ipants |
| Efficacy : cases were ascertained clinically and immunologically and outcomes in all cases are present- ed as scores at day 2 of follow up for "respiratory illness" and "constitutional illness" which does not include ILI symptoms (Figures 1 and 2). Adverse effects reported in Table 2 are all local and due to the aerosol. We only included nasal burning as the most significant |
| |

Amantadine and rimantadine for influenza A in adults (Review)

| Hayden 1980 (Continued) | | ta on persistence and shedding of influenza A viruses from the upper airways. Vi- ntly lower in the treatment arm |
|--|-----------------------|---|
| Notes | assessment impossible | ndomised, but no description of allocation and concealment is given making its e. Additionally the rationale for distinguishing between constitutional and respi- r, results of outcomes are not clearly reported (mean scores only are given) and are not explained |
| Risk of bias | | |
| Bias | Authors' judgement | Support for judgement |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

Hayden 1981

| Methods | A toxicity study reporting a randomised controlled trial undertaken in an unspecified period in USA an published in 1981 | | |
|---------------|---|--|--|
| Participants | The setting is that of a state farm insurance company and the participants were 251 adult volunteers, aged between 18 to 65 (mean age of 32) | | |
| Interventions | Two trials were carried out simultaneously, both involving rimantadine and amantadine. One was a low dose (200 mg daily of each drug, n = 52) and the other a higher dose trial (300 mg daily of each drug, n = 199). The low dose trial, however, has been excluded due to the absence of any 'cases' data, and the lack of outcomes | | |
| Outcomes | Safety: systemic symptoms only with no other classification were noted, although not specified | | |
| Notes | The practices of randomisation, allocation and concealment are not further defined, although all doses were stated as being administered by a project nurse. This is a poorly reported trial as no detailed clas- sification of adverse effects is given, which is a strange practice for a toxicity study. Additionally, data reported in the text are not consistent with that in Table 1.c. Overall 41 out of 67 (61%) in the amanta- dine arm, 13 out of 66 in the placebo arm (20%) and 18 out of 63 in the rimantadine arm (29%) experi- enced adverse effects | | |
| Risk of bias | | | |

| Bias | Authors' judgement | Support for judgement |
|--|--------------------|-----------------------|
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

Hayden 1986

| Methods | Randomised double-blind, placebo controlled treatment trial of oral rimantadine The trial took place in the universities of Virginia and Michigan in 1983 | | |
|---------------|---|--|--|
| Participants | 14 adults with confirmed A/Bangkok/1/79(H3N2) influenza | | |
| Interventions | Participants were randomised to receive either rimantadine 200 mg once daily (mean age 28 years) or placebo (mean age 23 years) for 5 days. Treatment started within 48 hours of symptom onset | | |

Amantadine and rimantadine for influenza A in adults (Review)



Hayden 1986 (Continued)

 Outcomes
 Efficacy: nasal virus shedding, duration of fever (in hours) and symptom scores (presented broken down into systemic – headache, chills, malaise, etc. and respiratory). Average duration of fever in the rimantadine arm was 31 hours (SD 22 hours) and 68 (SD 8 hours) in the placebo group

 Notes
 Although the trial is extremely clearly reported, no description of allocation and concealment is given

 Risk of bias Bias

 Authors' judgement
 Support for judgement

 Allocation concealment (selection bias)
 Unclear risk

Hornick 1969a

| Placebo controlled, double-blind treatment comparison of amantadine 100 mg with lactose placebo twice daily for 10 days. 94 inmates were randomised to receive amantadine and 103 placebo in January 1968, during an epidemic of influenza A2 Participants were 153 inmates of 4 prisons: Jessup, Richmond, Walls and Wynne. Hornick 1969a reports results from the Jessup site (renamed Jessup/Maryland) Amantadine n = 15 mean duration 66 hours, placebo n = 15, duration 92 hours, duration SD = 35 hours (for both arms). We transformed the duration data into 24-hour days |
|---|
| Amantadine n = 15 mean duration 66 hours, placebo n = 15, duration 92 hours, duration SD = 35 hours |
| |
| |
| Efficacy: influenza diagnosis was made on the basis of clinical and laboratory findings. The study re- ported data on persistence and shedding of influenza A viruses from the upper airways |
| The word "randomised" in not visible in the text, however denominators in each of the arms are highly suggestive of randomisation. No mention of the allocation procedure is made in the text, nor are dropouts mentioned. |
| Overall results show that participants could be divided into "rapid resolvers" to treatment (whose ill- ness resolved within 36 hours or less), medium resolvers (whose illness resolved within 24 to 36 hours) and slow resolvers (whose illness resolved in more than 36 hours) in both arms of the trial |
| |

| Bias | Authors' judgement | Support for judgement |
|--|--------------------|-----------------------|
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

| Hornick 1969b | |
|---------------|---|
| Methods | Placebo controlled, double-blind treatment comparison of amantadine 100 mg with lactose placebo twice daily for 10 days. 94 inmates were randomised to receive amantadine and 103 placebo in January 1968, during an epidemic of influenza A2 |
| Participants | Richmond site (renamed Hornick/Richmond) |
| Interventions | Amantadine n = 21, mean duration 60.9 hours, placebo n = 28, mean duration 80.1 hours, duration SD = 33 hours (for both arms) |

Amantadine and rimantadine for influenza A in adults (Review)

| Hornick 1969b (Continued) | | | |
|--|---|-----------------------|--|
| Outcomes | Efficacy: influenza diagnosis was made on the basis of clinical and laboratory findings. The study re- ported data on persistence and shedding of influenza A viruses from the upper airways | | |
| Notes | The word "randomised" in not visible in the text, however denominators in each of the arms are highly suggestive of randomisation. No mention of the allocation procedure is made in the text, nor are drop- outs mentioned. | | |
| | Overall results show that participants could be divided into "rapid resolvers" to treatment (whose ill- ness resolved within 36 hours or less), medium resolvers (whose illness resolved within 24 to 36 hours) and slow resolvers (whose illness resolved in more than 36 hours) in both arms of the trial | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear | |

Hornick 1969c

| Methods | Placebo controlled, double-blind treatment comparison of amantadine 100 mg with lactose placebo twice daily for 10 days. 94 inmates were randomised to receive amantadine and 103 placebo in Januar 1968, during an epidemic of influenza A2 |
|---------------|---|
| Participants | Walls site (renamed Hornick/Walls) |
| Interventions | Amantadine n = 23 mean duration 65.1 hours, placebo n = 20, mean duration 88.3 hours, duration SD = 28 hours (for both arms) |
| Outcomes | Efficacy: influenza diagnosis was made on the basis of clinical and laboratory findings. The study re- ported data on persistence and shedding of influenza A viruses from the upper airways |
| Notes | The word "randomised" in not visible in the text, however denominators in each of the arms are highly suggestive of randomisation. No mention of the allocation procedure is made in the text, nor are drop- outs mentioned. |
| | Overall results show that participants could be divided into "rapid resolvers" to treatment (whose ill- ness resolved within 36 hours or less), medium resolvers (whose illness resolved within 24 to 36 hours) and slow resolvers (whose illness resolved in more than 36 hours) in both arms of the trial |

| Bias | Authors' judgement | Support for judgement |
|--|--------------------|-----------------------|
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

Hornick 1969d Methods Placebo controlled, do twice daily for 10 days.

Placebo controlled, double-blind treatment comparison of amantadine 100 mg with lactose placebo twice daily for 10 days. 94 inmates were randomised to receive amantadine and 103 placebo in January 1968, during an epidemic of influenza A2

Amantadine and rimantadine for influenza A in adults (Review)

Hornick 1969d (Continued)

| Bias | Authors' judgement Support for judgement | | |
|---------------|---|--|--|
| Risk of bias | | | |
| | Overall results show that participants could be divided into "rapid resolvers" to treatment (whose ill- ness resolved within 36 hours or less), medium resolvers (whose illness resolved within 24 to 36 hours) and slow resolvers (whose illness resolved in more than 36 hours) in both arms of the trial | | |
| Notes | The word "randomised" in not visible in the text, however denominators in each of the arms are highly suggestive of randomisation. No mention of the allocation procedure is made in the text, nor are dropouts mentioned | | |
| Outcomes | Efficacy: influenza diagnosis was made on the basis of clinical and laboratory findings. The study re- ported data on persistence and shedding of influenza A viruses from the upper airways | | |
| Interventions | Amantadine n = 17, mean duration 49.8 hours, placebo n = 17, mean duration 82.1 hours, duration SD = 39 hours (for both arms) | | |
| Participants | Wynne site (renamed Hornick/Wynne) | | |

| to 2000 | | |
|--|--|-----------------------|
| Methods | Controlled treatment trial of the efficacy of amantadine during the 1999 A/Sidney/05/97 influenza sea- son. The trial was carried out in Japan among clinic attenders within 48 hours of testing positive for in- fluenza. Allocation is described only as semi-randomised. Follow up is not described | |
| Participants | 49 people aged 35.6 (mean) with influenza took part No drop-outs are mentioned | |
| Interventions | Participants were assigned to either oral amantadine 100 mg daily and standard medication (combina- tion of antibiotics, NSAIDs, antihistamines and cough mixtures) or standard medication only for 4.2 + 1.2 days | |
| Outcomes | Efficacy: maximum body temperature (in degrees C) after treatment in days Duration of fever > 37 °C and > 38 in days Duration of aching in days Duration of fatigue in days | |
| Notes | The authors conclude that amantadine hastens significantly the resolution of fever > 38 °C by 0.7 a day Assessment of the trial was hampered by the limited information available in English | |
| Risk of bias | | |
| Bias | Authors' judgement | Support for judgement |
| Allocation concealment (selection bias) | High risk | C - Inadequate |
| | | |

| Cantor 1980 | | | |
|--|--|-----------------------|--|
| Methods | Prophylaxis, double-blind, randomised controlled trial of the efficacy and safety of oral amantadine compared to a (not further defined) placebo. The trial took place over the period 20 February to 7 March 1978 in the military barracks at Fort Sam Houston (FSH), Texas and the target serotype was A/ USSR/77 | | |
| Participants | Trial participants were 139 healthy paramedic recruits (mean age 22 years) | | |
| Interventions | Participants were randomised to receive either amantadine 100 mg tablets twice daily (n = 64) or place- bo (n = 62) | | |
| Outcomes | Laboratory: paired sera were obtained at the beginning of the study and 5 weeks later | | |
| | Efficacy: case definition consisted of a list of recognised symptoms by severity, with serological confir- mation and adverse effects were recorded in the questionnaires | | |
| Notes | Although the trial was randomised and allocation concealment was protected through a system of sealed envelopes, the difficulty in reconciling figures and understanding what actually happened during the trial makes this a very poorly reported study. No reason for the loss of 20 participants to follow up is given | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Low risk | A - Adequate | |

| Kitamoto | 1968 |
|-----------|------|
| Ritaniutu | T200 |

| Methods | Randomised, double-blind, placebo controlled treatment trial of oral amantadine. The trial took place during the 1967 to 1968 influenza season throughout Japan when A2/Kumamoto/1/65 virus and its drifted variants were circulating in the community | | | |
|--|--|--|--|--|
| Participants | 355 participants | | | |
| Interventions | Amantadine (n = 182) (toms | Amantadine (n = 182) (200 mg/day for adults) or placebo (n = 173) within 48 hours of developing symptoms | | |
| Outcomes | Efficacy: case definition was based on symptoms and serological confirmation | | | |
| Notes | Although the trial is adequately reported and clearly randomised, no description of allocation and con- cealment is given making its assessment impossible | | | |
| Risk of bias | | | | |
| Bias | Authors' judgement | Support for judgement | | |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear | | |



Kitamoto 1971

| Methods | Randomised, double-blind, placebo controlled treatment trial of oral amantadine. The trial took place during the 1968 to 1969 influenza season throughout Japan | | |
|--|--|------------------------|--|
| Participants | 79 participants were randomised to receive amantadine (200 mg/day for adults) and 76 placebo for 7 days within 48 hours of developing symptoms | | |
| Interventions | Amantadine and place | po in a treatment role | |
| Outcomes | Case definition was based on symptoms and serological confirmation of infection. Outcomes were du- ration of fever (in days), defined as a temperature of more that 37 degrees centigrade and there were no drop-outs from the study Safety: adverse events were reported as nil | | |
| Notes | Although randomisation was clearly applied and no drop-outs are reported, no description of allo- cation and concealment is given making its assessment impossible. In addition the follow up of 353 participants is unclear. The authors conclude that treatment started within 2 days of symptom onset shortened significantly the duration of illness and of high fever but did not affect asymptomatic in- fluenza | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear | |

Knight 1970

| - | | | | |
|--|---|--|--|--|
| Methods | Randomised, double-blind treatment trial comparison, took place during a A2/Hong Kong influenza pandemic in January 1969 | | | |
| Participants | 29 unvaccinated male inmates aged 22 to 42 years of 2 units of the Texas Department of Correc Only participants with both serological and culture (nasal shedding of the virus) confirmation c took part: of the 37 original participants only 29 were included | | | |
| | | Illness had become manifest an average of 42 hours before for amantadine-treated patient and 36 hours for placebo (the authors report that the differences were not significant) | | |
| Interventions | Amantadine compared to placebo. Participants were randomised to receive either amantadine 100 mg (n = 13) or placebo capsules (n = 16) twice daily for at least 6 days | | | |
| Outcomes | Efficacy: outcomes were clinical (duration of fever and symptom score) or serological and laborato- ry-based (antibody titre and viral shedding). Illness had become manifest an average of 42 hours before for amantadine-treated patient and 36 hours for placebo (the authors report that the differences were not significant) | | | |
| Notes | Although randomisation was clearly applied and no drop-outs are reported, no description of alloca- tion and concealment is given making its assessment impossible | | | |
| Risk of bias | | | | |
| Bias | Authors' judgement | Support for judgement | | |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear | | |

Amantadine and rimantadine for influenza A in adults (Review)



Millet 1982

| Allocation concealment (selection bias) | Low risk | A - Adequate | |
|---|---|--|--|
| Bias | Authors' judgement | Support for judgement | |
| Risk of bias | | | |
| Notes | The study was randomised and certainly double-blind, with centralised preparation of active and placebo tablets | | |
| Outcomes | Safety: subjective side effects (from a predefined list) with a grading of 1 (mild) to 3 (severe) and perfor- mance testing | | |
| Interventions | Amantadine 100 mg and chlorpheniramine placebo (n = 10), amantadine 100 mg and chlorpheniramine 4 mg (n = 11), chlorpheniramine 4 mg and antiviral placebo (n = 11), rimantadine 100 mg and chlor-pheniramine placebo (n = 10) or antiviral placebo and chlorpheniramine placebo (n = 10) twice daily for 3 to 4 days | | |
| Participants | 52 healthy adult volun | teers recruited from the University of California at Los Angeles (USA) | |
| Methods | Randomised, double-blind safety trial of amantadine alone or in combination with chlorpheniramine (an anti-histaminic compound), rimantadine alone or chlorpheniramine. The trial was conducted be- tween October 1979 and January 1980 | | |

Monto 1979

| Methods | Prophylaxis, randomised, double-blind, placebo controlled trial carried out among healthy student vol- unteers at the University of Michigan, USA. The trial began in February 1978, during an outbreak of in- fluenza A/USSR-like (H1N1) and lasted 7 weeks (the when the outbreak was over as confirmed by clini- cal and serological surveillance) | | |
|--|--|--|--|
| Participants | 286 university students | s aged 18 to 24 | |
| Interventions | | Amantadine 100 mg twice daily or placebo. Participants were followed up weekly for 7 weeks, the re- maining duration of the outbreak of influenza | |
| Outcomes | Efficacy: the study reports separately the efficacy on asymptomatic cases of influenza (diagnosed from a rise in antibody titres). Respiratory and adverse effect symptoms were recorded in weekly calen- dar-diaries and paired sera samples were taken at the beginning and end of the trial (including 5 drop- outs from the trial because of clinical influenza or major adverse effects) from 281 of the participants | | |
| Notes | The trial is well written and appears well conducted with only minor discrepancies in total numbers. The practice of random allocation and concealment appear adequate. Analysis was not carried on an intention to treat basis | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Low risk | A - Adequate | |
| | | | |

| Methods | Cross-over, prophylaxis, controlled trial conducted in a school near Chicago (USA) between 2 December and 20 December 1968 (period 1). The study took place during the pandemic of A/Hong Kong/1/68 (H3N2) | | |
|--|---|--|--|
| Participants | 105 unvaccinated volunteer students aged around 21 years old took part | | |
| Interventions | At the beginning and the end of period 1 blood samples were taken from participants. During the 18 days of period 1 the 53 individuals in the intervention arm received 100 mg of amantadine hydrochlo- ride twice daily and the control arm (a not further described oral placebo). After a 'holiday' period of 16 days (period 2) the participants were crossed over to assess whether withdrawal of amantadine ren- dered participants more or less susceptible to infection. The procedure was inverted and students were followed up for a further 10 days (period 3) | | |
| Outcomes | Efficacy: ILI cases (from a symptom score coupled with virus-specific antibody rise). This allowed the authors to distinguish non-influenza related illness during periods 1 and 2 (reported in Table 1 without an indication of intensity of symptoms). During period 1 (Table 2) 6 students had influenza (5 with sero-logical confirmation) out of the 53 in the intervention arm and 10 (2 with serological confirmation) out of the 53 in the intervention arm and 10 view of the serological confirmation) out of the 50 in the intervention arm and 10 view of the serological confirmation). | | |
| | Adverse events are described as insignificant but no data are reported | | |
| Notes | The authors conclude that amantadine is effective in a preventive role. The practices of randomisation, allocation and concealment are not defined, making it impossible to assess methodological rigour, al- though the similar numbers of participants in each arm led us to consider this a randomised trial. No drop-outs are reported | | |
| Risk of bias | | | |
| Bias | Authors' judgement Support for judgement | | |
| Allocation concealment (selection bias) | Unclear risk B - Unclear | | |
| Náté 1970 | | | |
| Methods | Cluster randomised, placebo controlled prevention and treatment trial of amantadine during the 1969 Hong Kong influenza epidemic in 7 military units in Hungary. Treatment was started as soon as the out- break was recognised and lasted for up to 3 weeks. Randomisation was carried out by dormitory within the same barrack block | | |
| Participants | 4740 healthy male soldiers aged 18 to 21 from seven different military units | | |
| Interventions | Participants were randomised to oral placebo or amantadine 100 mg twice daily as soon as the out- break became clear. However in the units 1 to 4 the allocation procedure worked well but in the re- maining units (5 to 7) the outbreak was only recognised later and some soldiers were partially treat- ed or not treated at all. Surveillance showed that in units 5 and 7 influenza A activity was minimal and mixed with that of other agents, whereas in units 1 to 4 and 6 influenza activity was high. Because of uneven exposure to amantadine, only data for units 1 to 4 were extracted | | |
| Outcomes | Serological/laboratory: viral isolation or antibody response Paired sera (from 25 subjects from each arm) | | |
| | Effectiveness: cases of influenza and cases of ILI (no clear definition was given) Fever (duration and peak) Admission to infirmary Complications | | |

Amantadine and rimantadine for influenza A in adults (Review)



| Máté 1970 (Continued) | |
|-----------------------|--|
| | In the results section outcomes relating to complications and duration of fever are reported by 3 sub- groups of soldiers admitted to hospital/infirmary during the trial. These groups are 'V' (amantadine re- cipients), 'U' (placebo recipients) and 'P' (recipients of no intervention). The complications data are re- ported in 'U' + 'P' groups with no breakdown, leading to loss of the data |
| | As the data relate to soldiers who are already admitted to the infirmary with influenza symptoms, this part of the trial is treatment. ILI outcome data are unclearly reported |
| | Safety: the text mentions a pre-trial test for safety on 50 soldiers in which no adverse effects were de- tected but no further data are reported |
| Notes | The authors conclude that amantadine failed to affect morbidity but reduced peak and duration of fever, duration of stay in infirmary and the incidence of complications, especially lower respiratory tract infections. Considering its age the trial is reported reasonably but lack of clarity over allocation schedules, outcome definition and amantadine coverage has led to loss of data. In practice the trial is a mix of prevention and treatment. The analysis presented in the report did not account for the clustering of allocation by dormitory, and inadequate information is provided to allow a post-analysis adjustment for clustering |
| Risk of bias | |
| | |

| Bias | Authors' judgement | Support for judgement |
|--|--------------------|-----------------------|
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

| fta | | |
|-----|--|--|

| Bias | Authors' judgement Support for judgement |
|---------------|---|
| Risk of bias | |
| | A well-reported trial. We assume all participants were healthy |
| Notes | The authors conclude that amantadine was highly effective in preventing influenza from Hong/Kong A virus and appeared to prevent symptom expression without interfering with the immune response. |
| | Safety: no adverse events are reported |
| | Effectiveness: cases of influenza Cases of ILI (influenza symptoms but no antibody rise) Antibody rise with no influenza symptoms |
| Outcomes | Serological/laboratory: antibody response with paired sera (fourfold or more increase was considered positive) |
| Interventions | Oral placebo or amantadine 100 mg twice daily for 20 days |
| Participants | 215 healthy subjects in the 4 trial locations. No withdrawals are reported. Data are reported for all par- ticipants by pre-treatment serological status and study site |
| Methods | Randomised, double-blind, placebo controlled prevention trial of amantadine during the 1969 Hong Kong influenza epidemic in 4 locations in Romania (2 sanatoria, 1 blood transfusion centre and 1 pub- lic health department). Treatment was started before influenza activity was recognised. Randomisatio is not described but both drugs and placebo were centrally prepared and were indistinguishable apart from a letter C for the active principle and R for placebo |

Amantadine and rimantadine for influenza A in adults (Review)



Nafta 1970 (Continued)

| Allocation concealment | Unclear risk | В |
|------------------------|--------------|---|
| (selection bias) | | |

B - Unclear

| Oker-Blom 1970 | | | |
|------------------------|---|--|--|
| Methods | Prophylaxis, randomised, double-blind, placebo controlled trial begun on 20 January 1969 in Helsi Finland and lasting for 30 days during the pandemic of influenza A2 Hong Kong | | |
| Participants | Participants were 391 healthy medical students (132 females and 259 males) aged on average 22 yea | | |
| Interventions | Participants were randomised to receive either amantadine 100 mg or placebo (tablets containing calcium lactate) twice daily. Additionally the authors analysed age-stratified results for each arm (al-though the results are not included in the text) | | |
| Outcomes | Efficacy: cases were defined on the basis of self-reported (by questionnaire) clinical symptoms and serological titres | | |
| | Safety: headache, diffi | culty in concentration, insomnia, vertigo, malaise | |
| Notes | The authors conclude that amantadine has a protective efficacy of 52%. Randomisation and allocation are not sufficiently described whereas the placebo tablets are described as "identical". No reason is given for the 2 drop-outs from the study. It is perhaps debatable whether some of the self-reported in-fluenza cases really were such | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment | Unclear risk | B - Unclear | |

Payler 1984

(selection bias)

| Risk of bias | | | |
|---------------|--|--|--|
| Notes | Although the trial was randomised, the design is open and the analysis was not carried out on an inten- tion-to-treat basis. Additionally, length of follow up is not reported | | |
| Outcomes | Boys were randomised to receive either amantadine 100 mg daily (n = 267 or no treatment n = 269) 14 days. The trial started on 18 February 1983, as soon as the cases of laboratory-confirmed influen were described Efficacy: case definition is unclear but from the results (Table 2) it would appear that clinical definit and laboratory isolates were used | | |
| Interventions | | | |
| Participants | Boys aged 13 to 19 years had been previously immunised in the autumn of 1982 (viral strain not report- ed) | | |
| Methods | Prophylaxis randomised open trial of 536 members of Malvern Boys School in the UK. A/H1N1 was t prevalent viral strain | | |

Amantadine and rimantadine for influenza A in adults (Review)



Payler 1984 (Continued)

| Allocation concealment | High risk |
|------------------------|-----------|
| (selection bias) | |

C - Inadequate

| Methods | Randomised trial of the effects of oral amantadine for 28 days. Individuals in each arm, 3 days after commencing treatment were assigned to be injected with either influenza vaccine (2059 individuals) or subcutaneous saline placebo (2124 individuals). Peckinpaugh/1 was a randomised, factorial, dou- ble-blind, controlled trial conducted between 5 December 1967 and 1 June 1968 during the 1968 to 1969 pandemic | | |
|--|--|-----------------------|--|
| Participants | 4183 US Naval trainees. Data for those who received vaccine were not extracted | | |
| Interventions | Comparison of the effects of oral amantadine (200 mg daily) on 2079 individuals and oral placebo on 2104 individuals with and without the addition of vaccine | | |
| Outcomes | Efficacy: outcomes are presented in histogram form (Figure 3) by intervention arm (amantadine and in- fluenza vaccine - 1039 individuals, influenza vaccine and oral placebo - 1020 individuals, amantadine and subcutaneous placebo - 1040 and subcutaneous and oral placebo - 1084 individuals). Outcomes in the figure are: % of participants who did not seek treatment, sought treatment as outpatients or were admitted to local hospital. Adverse events are described as nil | | |
| Notes | Both studies report no effect, probably because of low influenza circulation. The study was definite- ly randomised and arrangements to conceal allocation appear adequate. The standard of reporting is poor with outcomes reported in histogram form, making estimation of their incidence necessary. Additionally contradictory statements in the text (concerning the definition of cases and serological outcomes) make interpretation of the text difficult | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Low risk | A - Adequate | |

Peckinpaugh 1970b

| Methods | Peckinpaugh 2 reports the results of a study conducted from December 1968 to May 1969 during the 1968 to 1969 pandemic |
|---------------|--|
| Participants | US Naval recruits |
| Interventions | Recruits were randomised to receive either amantadine 200 mg daily (n = 1329) or placebo (n = 1321) orally for 20 days |
| Outcomes | As in Peckinpaugh 1, serological assessment was carried out on one fourth of participants (416 sera pairs). Outcome definition and presentation (Figure 6) were similar to those in Peckinpaugh 1, as is our transformation of them |
| Notes | Both studies report no effect, probably because of low influenza circulation. The study was definite- ly randomised and arrangements to conceal allocation appear adequate. The standard of reporting is poor with outcomes reported in histogram form, making estimation of their incidence necessary. |

Amantadine and rimantadine for influenza A in adults (Review)



Peckinpaugh 1970b (Continued)

Additionally contradictory statements in the text (concerning the definition of cases and serological outcomes) make interpretation of the text difficult

| Risk of bias | | | |
|--|--------------------|-----------------------|--|
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Low risk | A - Adequate | |

Pettersson 1980

| Prophylaxis, randomised, double-blind, placebo controlled trial in a military training centre in Finland commencing on 17 January 1978. At the time an epidemic of A H1N1 was underway in the surrounding community | | |
|---|---|--|
| 192 participants (military recruits aged 21 average) | | |
| Participants were randomly assigned to receive either amantadine 200 mg daily of calcium lactate placebo tablets | | |
| Efficacy: participants were followed up for 3 weeks and cases were defined on the basis of retrospective analysis of questionnaires and antibody titre rise in paired sera. Infection with influenza A H3N2 virus also was verified serologically | | |
| Although the study report is well written and random allocation was carried out on the basis of random number tables, allocation concealment was unspecified and intention-to-treat analysis did not take place | | |
| | | |
| Authors' judgement | Support for judgement | |
| Unclear risk | B - Unclear | |
| - | commencing on 17 Jan community 192 participants (milita Participants were rand placebo tablets Efficacy: participants w analysis of questionnai also was verified serold Although the study rep number tables, allocati place Authors' judgement | |

Plesnik 1977

| Methods | Prophylaxis, randomised, double-blind, controlled trial of amantadine during an epidemic of influenza A/Victoria/42/75 (H3N2) virus. The trial took place in the middle of March 1976 in the former Czechoslo- vakia and lasted 20 days |
|---------------|--|
| Participants | Participants were 1133 healthy students, industrial workers and elderly aged 10 to 69 |
| Interventions | Participants were randomised to receive either amantadine 100 mg once or twice daily (n = 574) or placebo (n = 559), 1 or 2 tablets daily |
| Outcomes | Efficacy: case definition was based on influenza symptoms and antibody titre rise from paired sera and viral isolation (although from the translated text it is not clear whether this practice covered the whole study population). Adverse effects are reported in detail in the trial |



Plesnik 1977 (Continued)

Notes

Although randomisation, allocation and analysis on an intention-to-treat basis are not described, concealment is described as a centralised scheme with coded tablets and is therefore assessed as adequate

Risk of bias

| Bias | Authors' judgement | Support for judgement | |
|--|--------------------|-----------------------|--|
| Allocation concealment (selection bias) | Low risk | A - Adequate | |

Quarles 1981

| C | | | |
|--|--|-----------------------|--|
| Methods | Prophylaxis randomised double blind placebo-controlled trial carried out at the University of Texas | | |
| Participants | Participants were 444 healthy students aged 18 to 24 who were followed up for 6 weeks on the basis weekly of self-reporting. The trial was commenced on week 7 (? mid-February) 1978 during an epidem- ic of influenza A/USSR/90/77 virus | | |
| Interventions | Students were randomised to receive either amantadine (100 mg), or rimantadine (100 mg), or placebo (not specified) tablets twice daily | | |
| Outcomes | Efficacy: cases were defined on the basis of paired sera and clinical symptoms. Viral isolation from throat swabs was additionally carried out | | |
| Notes | The practices of randomisation, allocation and concealment are not defined, making it impossible to assess methodological rigour. Analysis on the basis of intention-to-treat was not carried out and the overall quality of reporting is poor (with no complete details of reasons for drop-out). This study was probably commenced at of just after the peak of influenza transmission with a consequent possible underestimation of the efficacy of prophylaxis interventions (due to late commencement of prophylaxis) | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear | |
| | | | |

| Methods | Randomised, placebo controlled treatment trial of the efficacy of rimantadine during the 1967 A2/ Iowa/1/67 influenza season. The trial was carried out in the Iowa state penitentiary, USA. Treatment was started as soon as the outbreak was recognised and lasted 10 days. Randomisation is not de- scribed in detail but allocation concealment was centralised using coded bottles |
|--------------|---|
| Participants | 22 unvaccinated inmates of the penitentiary took part. 19 had a retrospective diagnosis of influenza in- fection (through antibody titre assessment), 1 dropped out of the trial of the placebo arm and no fur- ther details are given of the 3 participants who had ILI symptoms but no serological confirmation of in- fection. All participants were enrolled if they had ILI symptoms of less than 48 hours duration. All 18 in- fluenza cases had fever, chills, malaise and fatigue, nasal stuffiness, sore or dry throat and nasal mu- cosal swelling |

Rabinovich 1969 (Continued)

Cochrane Library

| Interventions | Participants were randomised to oral placebo or rimantadine 150 mg twice daily for 10 days. No symp- tomatic medication was given | | |
|--|--|-----------------|--|
| Outcomes | Serological/laboratory: viral isolation antibody response (paired sera) | | |
| | Effectiveness: cases of influenza (symptoms of ILI plus antibody response); duration of symptoms (fever - oral temp > 37 °C - nasal stuffiness, headache, sneezing, chills, sore throat, sweats, cough, malaise, sputum, myalgia, chest pain, eye pain, anorexia, apathetic appearance, hyperemic nasal mu cosa, nasal discharge, nasal mucosa swelling, pharyngitis); severity of symptoms (4-point scale 0 = n present, 1 = mild, 2 = moderate, 3 = severe, 4 = very severe) | | |
| | The text does not mention any safety outcome data | | |
| Notes | The authors conclude that rimantadine hastens significantly the resolution of symptom ence in 36 hours). A well-reported trial considering its age | ıs (50% differ- | |
| Risk of bias | | | |
| Bias | Authors' judgement Support for judgement | | |
| Allocation concealment (selection bias) | Low risk A - Adequate | | |
| | | | |
| Reuman 1989 | | <u> </u> | |
| Methods | Reuman is the report of a double-blind, prophylaxis, randomised controlled trial. The study was con- ducted in Cincinnati Christ Hospital, Ohio, USA during the 1985 to 1986 influenza season. Participants were followed up weekly for 6 weeks through symptom diaries and visits and all those reporting with influenza-like illness (pre-defined from a list of signs and symptoms) were seen on 2 consecutive days for examination and collection of nasal washouts | | |
| Participants | Participants were 476 healthy hospital personnel aged 18 to 55 | | |
| Interventions | There were 3 arms to the trial: | | |
| | a) 159 subjects randomly allocated to receive amantadine 100 mg/day b) 159 subjects randomly allocated to receive amantadine 200 mg/day c) 158 subjects randomly allocated to receive identical looking placebo capsules each day | | |
| | To test compliance with the schedule random blood test were carried out on 48 subjects (all results were consistent with compliance) | | |
| Outcomes | Laboratory: paired sera were collected at the beginning and end of the trial and tested for antibodies against A/Chile/1/83 (H1N1), A/Philippines/1/82 (H3N2) and B/USSR/100/83, the 3 prevalent viruses during the study period. Viral assays were performed from nasal washouts | | |
| | Efficacy: ILI and influenza. 5 participants were infected in the placebo group, 2 subjects in the 100 mg group and 1 in the 200 mg group. The number of participants with at least 1 (systemic) adverse effect was 49 in the placebo arm, 47 in the 100 mg arm and 71 in the 200 mg arm, indicating a clear dose-ad- verse effect relationship (Table 1) | | |
| Notes | The authors conclude that influenza activity was too low to be detected by the study (which was pow- ered for a 30% incidence). The study is of good quality with computer-generated randomisation, checks on allocation concealment and good blinding procedures. No drop-outs are reported | | |
| | | | |



Reuman 1989 (Continued)

| Bias | Authors' judgement | Support for judgement |
|---|--------------------|-----------------------|
| Allocation concealment (selection bias) | Low risk | A - Adequate |

Schapira 1971

| Methods | Keighly by one general | controlled trial of amantadine conducted in the Northern English village of practitioner during the pandemic of the winter of 1968 to 1969. Follow up was |
|--|---|---|
| Participants | was evenly matched at screened prior to comn second antibody titre e | from firms, schools, a newspaper, a bank and families. Participants' mean age 38 years (active arm) and 38.5 years (placebo arm). All participants' sera were nencement and those who complained of symptoms of influenza underwent a stimation against A2/Hong Kong/1/68, the prevalent strain at the time. As it is pants were already infected prior to commencement the authors point out that laxis and treatment |
| Interventions | | ated to receive either amantadine 100 mg every 12 hours (157 participants) or apsule (140 participants), identified only by a code |
| Outcomes | Laboratory: paired sera Efficacy: ILI or influenza Adverse effects are not reported | |
| Notes | The authors conclude t dine. The practices of ra methodological rigour, a randomised trial. Allo treatment arm suffered ly 8 and 15 cases respec unavailable for the 49 s | hat the trial provides "marginal" evidence of prophylactic effect of amanta- andomisation and allocation are not described, making it impossible to assess although the similar numbers of participants in each arm led us to consider this cation concealment appeared adequate. 49 out of the 157 volunteers in the l influenza symptoms compared to 39 out 140 on the placebo arm. However, on- ctively were serologically confirmed as influenza (Table 3). 12 paired sera were ymptomatic volunteers in the treatment arm and 3 paired sera were unavailable o not explain the reason for the drop-outs |
| Risk of bias | | |
| Bias | Authors' judgement | Support for judgement |
| Allocation concealment (selection bias) | Unclear risk | B - Unclear |

| Methods | Prophylaxis, randomised, double-blind, placebo controlled trial carried out in 8 engineering schools around Leningrad, in the former USSR. The design comprised an "internal" (day students) and "exter- nal" (i.e. boarders) do-nothing control group. Differential follow up is recorded between placebo and amantadine arms (30 days) and controls (12 days). The trial was carried out during the pandemic of 1968 to 1969 |
|--------------|--|
| Participants | Participants were all males, aged 18 to 30. The study commenced around January 1969 with staggered starts, according to the results of school-specific surveillance against A2/Hong Kong/68 virus |

Smorodintsev 1970 (Continued)

| Bias | Authors! judgement Support for judgement |
|---------------|---|
| Risk of bias | |
| Notes | The authors conclude that amantadine was effective in preventing cases. Although randomisation oc- curred and concealment is described in an adequate manner, an unknown number of participants were given amantadine syrup. As no reference to syrup placebo is made in the text (only to tablet form), robustness of blinding and concealment is questionable. Additionally, the numerous discrepancies in the text make this a poorly reported trial |
| Outcomes | Efficacy: case definition was based on a list of symptoms plus a four-fold rise in serum antibody titres to influenza A |
| Interventions | Participants were randomised to receive amantadine 100 mg daily (n = 5092) or placebo tablets (n = 3175) |

| Blas | Authors Judgement | Support for Judgement |
|--|-------------------|-----------------------|
| Allocation concealment (selection bias) | Low risk | A - Adequate |

van Voris 1981

| Methods | Randomised in a double-blind design | | |
|------------------------|--|-----------------------|--|
| Participants | This treatment trial was carried out in February 1978 in University of Rochester, USA on 45 students with laboratory confirmed influenza A/USSR/77 [H1N1] | | |
| Interventions | Participants received either amantadine 200 mg daily (n = 14) or rimantadine 200 mg daily (n = 14) or inert placebo capsules (n = 12) and followed up for 5 days | | |
| Outcomes | Efficacy: outcomes reported were symptom scores (Figure 1), duration of temperature (Figure 2 and in the text at page 1130). We used the outcome 'number of participants with temperatures more than 37.5 °C', 48 hours after commencing medication (a dichotomous outcome). Data on mean duration of fever were not given. The study reported data on persistence and shedding of influenza A viruses from the upper airways at day 2. Viral titres were significantly lower in the treatment arm | | |
| Notes | The trial is well-reported and definitely randomised. Allocation concealment is described and appears to be adequate | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment | Low risk | A - Adequate | |

Wendel 1966

(selection bias)

| Methods | Prophylaxis and treatment randomised, double-blind, placebo controlled trial in a prison community in Philadelphia, USA in January 1963. Follow up length was 10 days |
|--------------|--|
| Participants | Participants were prison inmates aged 17 to 54 exposed to naturally occurring influenza A2Japan305/57 virus |

Amantadine and rimantadine for influenza A in adults (Review)

Wendel 1966 (Continued)

| Interventions | Placebo was a lactose tablet. In the preventive trial 794 inmates were randomised to receive amanta- dine 200 mg daily (n = 439) or placebo (n = 355) | | |
|---|--|--|--|
| Outcomes | Efficacy: cases were defined both clinically and serologically | | |
| Notes | | orted although randomisation was not good (as confirmed by uneven numbers in in the amantadine arm) and allocation concealment is adequate | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Low risk | A - Adequate | |

| Methods | | lind, placebo controlled treatment trial which took place in the Virginia State ng an outbreak of influenza A2/Rockville/1/65 in January to February 1968 | |
|--|---|--|--|
| Participants | Participants were 95 inmates whose symptoms developed in the previous 24 hours | | |
| Interventions | 95 inmates were randomised to receive either amantadine 100 mg (n = 23), or rimantadine 150 mg (n = 24) or placebo lactose capsules (n = 48) twice daily. Medication was commenced within 24 hours of developing symptoms and continued for 10 days | | |
| | The final denominators | s for which data are provided in the report are: | |
| | Placebo n = 39 Amantadine n = 20 Rimantadine n = 20 | | |
| Outcomes | Efficacy: only data relating to participants with temperatures of 37.5 °C or more on study were analysed for the duration of fever outcome and 1 participant was later f rise at the second antibody titre estimation. We have included duration of fever as in our meta-analysis | | |
| | The study reported data on persistence and shedding of influenza A viruses from the upper airways at day 2 of follow up | | |
| Notes | The authors conclude that amantadine was effective in reducing symptoms. Randomisation and allo- cation concealment are detailed in the text with individual coded "vials" (which we assume to mean containers) system. Intention-to-treat analysis appears to have taken place on individuals who fulfilled clinical and serological criteria | | |
| Risk of bias | | | |
| Bias | Authors' judgement | Support for judgement | |
| Allocation concealment (selection bias) | Low risk | A - Adequate | |



Younkin 1983

| break of influenza A/Brazil/78 (H1N1) in the winter of 1981ParticipantsThe 48 participants in the trial were otherwise healthy college students aged 17 to 20 from the L ty of Rochester, New York, USA. Participants all had ILI symptoms and some had (later) laborate firmed diagnosis of influenzaInterventionsParticipants were randomised to receive either amantadine 100 mg a day (n = 16), 200 mg a day or aspirin 3.25 grams a day (n = 17)OutcomesEfficacy: outcomes are presented as both symptoms scores (Figure 1) and duration of fever (in the The study reported data on persistence and shedding of influenza A viruses from the upper air day 4 of follow up. There were lower viral titres in the control groupNotesRandomisation and allocation concealment are well detailed in the text and concealment in pa appears very robust with an elaborate system of envelopes. To uniformly make up the tablets m | | | | |
|---|---------------|---|-----------------------|--|
| ty of Rochester, New York, USA. Participants all had ILI symptoms and some had (later) laborate firmed diagnosis of influenzaInterventionsParticipants were randomised to receive either amantadine 100 mg a day (n = 16), 200 mg a day or aspirin 3.25 grams a day (n = 17)OutcomesEfficacy: outcomes are presented as both symptoms scores (Figure 1) and duration of fever (in t The study reported data on persistence and shedding of influenza A viruses from the upper airv day 4 of follow up. There were lower viral titres in the control groupNotesRandomisation and allocation concealment are well detailed in the text and concealment in pa appears very robust with an elaborate system of envelopes. To uniformly make up the tablets n the aspirin arm had 2 placebo tablets included. Compliance was equally tested by daily questic Bias Authors' judgementSupport for judgementAllocation concealmentLow riskA - Adequate | Methods | Randomised, double-blind comparison of amantadine with aspirin. The trial took place during an out- break of influenza A/Brazil/78 (H1N1) in the winter of 1981 | | |
| or aspirin 3.25 grams a day (n = 17) Outcomes Efficacy: outcomes are presented as both symptoms scores (Figure 1) and duration of fever (in the study reported data on persistence and shedding of influenza A viruses from the upper airwiday 4 of follow up. There were lower viral titres in the control group Notes Randomisation and allocation concealment are well detailed in the text and concealment in parappears very robust with an elaborate system of envelopes. To uniformly make up the tablets in the aspirin arm had 2 placebo tablets included. Compliance was equally tested by daily question Risk of bias Authors' judgement Support for judgement Allocation concealment Low risk A - Adequate | Participants | The 48 participants in the trial were otherwise healthy college students aged 17 to 20 from the Universi- ty of Rochester, New York, USA. Participants all had ILI symptoms and some had (later) laboratory con- firmed diagnosis of influenza | | |
| The study reported data on persistence and shedding of influenza A viruses from the upper airviday 4 of follow up. There were lower viral titres in the control group Notes Randomisation and allocation concealment are well detailed in the text and concealment in parappears very robust with an elaborate system of envelopes. To uniformly make up the tablets in the aspirin arm had 2 placebo tablets included. Compliance was equally tested by daily question Risk of bias Authors' judgement Support for judgement Authors' judgement Allocation concealment Low risk | Interventions | Participants were randomised to receive either amantadine 100 mg a day (n = 16), 200 mg a day (n = 14) or aspirin 3.25 grams a day (n = 17) | | |
| appears very robust with an elaborate system of envelopes. To uniformly make up the tablets in the aspirin arm had 2 placebo tablets included. Compliance was equally tested by daily question Risk of bias Bias Authors' judgement Allocation concealment Low risk A - Adequate | Outcomes | Efficacy: outcomes are presented as both symptoms scores (Figure 1) and duration of fever (in the text). The study reported data on persistence and shedding of influenza A viruses from the upper airways at day 4 of follow up. There were lower viral titres in the control group | | |
| Bias Authors' judgement Support for judgement Allocation concealment Low risk A - Adequate | Notes | Randomisation and allocation concealment are well detailed in the text and concealment in particular appears very robust with an elaborate system of envelopes. To uniformly make up the tablets number, the aspirin arm had 2 placebo tablets included. Compliance was equally tested by daily questioning | | |
| Allocation concealment Low risk A - Adequate | Risk of bias | | | |
| | Bias | Authors' judgement | Support for judgement | |
| | | Low risk | A - Adequate | |

ILI = influenza-like illness

Characteristics of excluded studies [ordered by study ID]

| Study | Reason for exclusion | |
|----------------|---|--|
| Aoki 1986 | Contains insufficient data | |
| Atmar 1990 | This study reports the results of safety and pharmacokinetic studies of inhaled rimantadine aerosol in both healthy and ill (with confirmed influenza A and B) volunteers. Although the text reports that inhaled rimantadine appears efficacious against influenza, no data is presented and the text goes on to qualify the effectiveness as not statistically significant against placebo | |
| Baker 1969 | Insufficient data presented | |
| Bricaire 1990 | 45% of participants are children and no breakdown of data by age group is reported | |
| Bryson 1980 | Insufficient data presented | |
| Clover 1986 | Trial of the preventive effect of rimantadine in children within familles. Adults were not exposed to rimantadine or placebo | |
| Crawford 1988 | Ages of participants were outside protocol age range (majority of participants are aged under 15 years) | |
| Dawkins 1968 | RCT - experimental influenza trial | |
| Finklea 1967 | RCT - study population are children | |
| Galbraith 1969 | 40% of trial population are outside the protocol's age inclusion criteria. | |

Amantadine and rimantadine for influenza A in adults (Review)

| Study | Reason for exclusion |
|--------------------|---|
| Galbraith 1973 | Population is outside (target) age range. Does not fulfil study inclusion criteria |
| Hayden 1983 | Toxicity study presenting further analysis of the data contained in Hayden 1981 |
| Hayden 1985 | Cross-over study of the pharmacokinetics of amantadine and rimantadine in 12 adults. There are no clinical or safety outcomes presented |
| Hayden 1989 | This trial assesses efficacy, rate of transmission and resistance to rimantadine from index cases to contacts within families. The trial is excluded because 42% of contacts in families with serologically documented influenza A are children (Table 1) |
| Hayden 1991 | The study is a treatment RCT of both families and 12 adults. No age breakdown is given for the out- come data |
| Knight 1969 | RCT - insufficient data reported |
| O'Donoghue 1971 | Semi-randomised allocation. In part, patients were admitted to hospital, therefore do not fit in in- clusion criteria for 'healthy adults'. Additionally, patient ages are outside those stipulated in inclu- sion criteria |
| Rose 1980 | Non-randomised, non-placebo controlled study |
| Smorodinstev 1970a | Presents duplicate data of included study |
| Smorodintsev 1970b | Trial of preventive efficacy in artificially induced influenza |
| Togo 1968 | RCT - experimental influenza trial |
| Togo 1970 | Duplicate of Hornick (identical sample sizes, same sites, same period, not cross-referenced) |

RCT = randomised controlled trial

DATA AND ANALYSES

Comparison 1. Oral amantadine versus placebo (prophylaxis)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|-----------------------------|-------------------|-----------------------------|----------------------------------|-------------------|
| 1 Influenza cases | 11 | 4645 | Risk Ratio (M-H, Random, 95% CI) | 0.39 [0.24, 0.65] |
| 1.1 Unvaccinated population | 10 | 4109 | Risk Ratio (M-H, Random, 95% CI) | 0.45 [0.28, 0.74] |
| 1.2 Vaccinated population | 1 | 536 | Risk Ratio (M-H, Random, 95% CI) | 0.10 [0.03, 0.34] |
| 2 ILI cases | 14 | 17496 | Risk Ratio (M-H, Random, 95% CI) | 0.75 [0.64, 0.87] |
| 2.1 Unvaccinated population | 13 | 14901 | Risk Ratio (M-H, Random, 95% CI) | 0.75 [0.62, 0.90] |
| 2.2 Vaccinated population | 2 | 2595 | Risk Ratio (M-H, Random, 95% CI) | 0.42 [0.07, 2.52] |

Amantadine and rimantadine for influenza A in adults (Review)



Cochrane Database of Systematic Reviews

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|---|-------------------|-----------------------------|-------------------------------------|-------------------|
| 3 Adverse effects | 13 | | Odds Ratio (M-H, Random, 95% CI) | Subtotals only |
| 3.1 Gastrointestinal | 5 | 3336 | Odds Ratio (M-H, Random, 95% CI) | 2.56 [1.37, 4.79] |
| 3.2 Increased CNS activity | 9 | 5002 | Odds Ratio (M-H, Random, 95% CI) | 2.54 [1.50, 4.31] |
| 3.3 Decreased CNS activity | 7 | 3797 | Odds Ratio (M-H, Random, 95% CI) | 1.73 [0.86, 3.45] |
| 3.4 Dermatological changes | 3 | 918 | Odds Ratio (M-H, Random, 95% CI) | 1.55 [0.39, 6.20] |
| 3.5 All adverse effects | 6 | 4274 | Odds Ratio (M-H, Random, 95% CI) | 1.70 [0.99, 2.93] |
| 3.6 Withdrawals due to adverse effects | 6 | 2276 | Odds Ratio (M-H, Random, 95% CI) | 2.54 [1.60, 4.06] |
| 4 Viral nasal shedding or persistence in upper airways at 2 to 5 days | 1 | 79 | Risk Ratio (M-H, Random, 95% CI) | 0.68 [0.53, 0.87] |
| 5 Influenza cases (asymptomatic) | 4 | 963 | Risk Ratio (M-H, Random, 95% CI) | 0.85 [0.40, 1.80] |

Analysis 1.1. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 1 Influenza cases.

| Study or subgroup | Amantadine | Placebo | Risk Ratio | Weight | Risk Ratio |
|---|--|-----------------|---------------------|--------------------------------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI |
| 1.1.1 Unvaccinated population | on | | | | |
| Dolin 1982 | 2/113 | 27/132 | _ | 6.72% | 0.09[0.02,0.36] |
| Kantor 1980 | 9/59 | 9/51 | | 10.18% | 0.86[0.37,2.01] |
| Monto 1979 | 8/136 | 28/139 | | 10.84% | 0.29[0.14,0.62] |
| Muldoon 1976 | 1/53 | 8/52 | | 4.27% | 0.12[0.02,0.95] |
| Máté 1970 | 57/996 | 41/996 | + | 13.11% | 1.39[0.94,2.06] |
| Nafta 1970 | 0/112 | 13/103 | | 2.63% | 0.03[0,0.57] |
| Oker-Blom 1970 | 16/141 | 41/152 | -+- | 12.3% | 0.42[0.25,0.71] |
| Pettersson 1980 | 32/95 | 59/97 | + | 13.44% | 0.55[0.4,0.77] |
| Quarles 1981 | 15/107 | 20/99 | -+ | 11.78% | 0.69[0.38,1.28] |
| Reuman 1989 | 3/317 | 5/159 | + | 6.7% | 0.3[0.07,1.24] |
| Subtotal (95% CI) | 2129 | 1980 | • | 91.97% | 0.45[0.28,0.74] |
| Total events: 143 (Amantadine |), 251 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.38; Chi ² | =39.29, df=9(P<0.0001); I ² = | 77.09% | | | |
| Test for overall effect: Z=3.19(P | P=0) | | | | |
| 1.1.2 Vaccinated population | | | | | |
| Payler 1984 | 3/267 | 29/269 | + | 8.03% | 0.1[0.03,0.34] |
| Subtotal (95% CI) | 267 | 269 | → | 8.03% | 0.1[0.03,0.34] |
| | Fav | ours amantadine | 0.002 0.1 1 10 | ⁵⁰⁰ Favours placebo | |

Amantadine and rimantadine for influenza A in adults (Review)



| Study or subgroup | Amantadine | Placebo | | Ris | k Rat | io | | Weight | Risk Ratio |
|---|---|-----------------|-------|---------------------|-------|----|-----|-----------------|-------------------|
| | n/N n/N M-H, Random, 95% Cl | | | M-H, Random, 95% CI | | | | | |
| Total events: 3 (Amantadine), 2 | 9 (Placebo) | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | |
| Test for overall effect: Z=3.77(P | =0) | | | | | | | | |
| Total (95% CI) | 2396 | 2249 | | • | | | | 100% | 0.39[0.24,0.65] |
| Total events: 146 (Amantadine) | , 280 (Placebo) | | | | | | | | |
| Heterogeneity: Tau ² =0.47; Chi ² = | =48.95, df=10(P<0.0001); I ² | =79.57% | | | | | | | |
| Test for overall effect: Z=3.65(P | =0) | | | | | | | | |
| Test for subgroup differences: C | Chi ² =5.1, df=1 (P=0.02), l ² =3 | 80.41% | 1 | 1 | | 1 | | | |
| | Fav | ours amantadine | 0.002 | 0.1 | 1 | 10 | 500 | Favours placebo | |

Analysis 1.2. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 2 ILI cases.

| Study or subgroup | Amantadine | Placebo | Risk Ratio | Weight | Risk Ratio |
|---|---|---------------------|---------------------|--------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI |
| 1.2.1 Unvaccinated population | n | | | | |
| Callmander 1968 | 33/47 | 31/47 | + | 7.81% | 1.06[0.81,1.4] |
| Dolin 1982 | 10/113 | 54/132 | _ | 3.87% | 0.22[0.12,0.4] |
| Monto 1979 | 18/136 | 45/139 | -+- | 5.08% | 0.41[0.25,0.67] |
| Nafta 1970 | 2/112 | 19/103 | | 1.06% | 0.1[0.02,0.41] |
| Oker-Blom 1970 | 62/141 | 88/152 | + | 8.46% | 0.76[0.6,0.96] |
| Peckinpaugh 1970a | 634/1040 | 575/1084 | + | 10.15% | 1.15[1.07,1.24] |
| Peckinpaugh 1970b | 744/1329 | 766/1321 | + | 10.19% | 0.97[0.9,1.03] |
| Pettersson 1980 | 66/95 | 69/97 | + | 9.07% | 0.98[0.81,1.17] |
| Plesnik 1977 | 48/574 | 57/559 | -+- | 6.59% | 0.82[0.57,1.18] |
| Quarles 1981 | 42/107 | 44/99 | -+- | 7.18% | 0.88[0.64,1.22] |
| Schapira 1971 | 49/157 | 39/140 | -+- | 6.76% | 1.12[0.79,1.6] |
| Smorodintsev 1970 | 156/3885 | 195/2498 | + | 8.8% | 0.51[0.42,0.63] |
| Wendel 1966 | 5/439 | 15/355 | | 1.95% | 0.27[0.1,0.73] |
| Subtotal (95% CI) | 8175 | 6726 | • | 86.97% | 0.75[0.62,0.9] |
| Total events: 1869 (Amantadin | e), 1997 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.08; Chi ² | =122.45, df=12(P<0.0001); I | ² =90.2% | | | |
| Test for overall effect: Z=3.11(P | =0) | | | | |
| 1.2.2 Vaccinated population | | | | | |
| Payler 1984 | 7/267 | 42/269 | + | 2.86% | 0.17[0.08,0.37] |
| Peckinpaugh 1970a | 613/1039 | 612/1020 | + | 10.16% | 0.98[0.92,1.06] |
| Subtotal (95% CI) | 1306 | 1289 | | 13.03% | 0.42[0.07,2.52] |
| Total events: 620 (Amantadine |), 654 (Placebo) | | | | |
| Heterogeneity: Tau ² =1.58; Chi ² | =20.65, df=1(P<0.0001); l ² = | 95.16% | | | |
| Test for overall effect: Z=0.94(P | =0.35) | | | | |
| Total (95% CI) | 9481 | 8015 | • | 100% | 0.75[0.64,0.87] |
| Total events: 2489 (Amantadin | e), 2651 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.06; Chi ² | =142.92, df=14(P<0.0001); I | ² =90.2% | | | |
| Test for overall effect: Z=3.66(P | =0) | | | | |
| Test for subgroup differences: | Chi ² =0.38. df=1 (P=0.54). I ² = | =0% | ĺ | | |

| Study or subgroup | Amantadine | Placebo | Odds Ratio | Weight | Odds Ratio |
|--|---|---------|-------------------------------------|------------------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI |
| 1.3.1 Gastrointestinal Callmander 1968 | 0/40 | 4/49 | | 10.020/ | 2 64[0 72 8 0] |
| | 9/48 | 4/48 | | 16.02% 12.17% | 2.54[0.72,8.9] |
| Hayden 1981 Plesnik 1977 | 13/67 | 2/66 | | 12.17% | 7.7[1.66,35.65] |
| Reuman 1989 | 15/387 | 3/419 | | | 5.59[1.61,19.47] |
| | 29/317 | 12/159 | | 28.71% | 1.23[0.61,2.49] |
| Smorodintsev 1970 | 43/1313 | 8/512 | | 26.97% | 2.13[1,4.57] |
| Subtotal (95% CI) | 2132 | 1204 | | 100% | 2.56[1.37,4.79] |
| Total events: 109 (Amantadine), 2 | | F0/ | | | |
| Heterogeneity: Tau ² =0.23; Chi ² =7. Test for overall effect: Z=2.94(P=0 | | 5% | | | |
| | 1 | | | | |
| 1.3.2 Increased CNS activity | | | | | |
| Callmander 1968 | 8/48 | 2/48 | + | 7.1% | 4.6[0.92,22.93] |
| Dolin 1982 | 19/145 | 6/148 | + | 12.42% | 3.57[1.38,9.22] |
| Hayden 1981 | 26/67 | 5/66 | │ • | 11.54% | 7.74[2.75,21.79] |
| Monto 1979 | 50/144 | 40/142 | - + | 17.48% | 1.36[0.82,2.24] |
| Oker-Blom 1970 | 18/141 | 6/152 | + | 12.36% | 3.56[1.37,9.25] |
| Plesnik 1977 | 16/387 | 5/419 | — • — · | 11.75% | 3.57[1.3,9.84] |
| Reuman 1989 | 70/317 | 25/159 | + | 17.47% | 1.52[0.92,2.51] |
| Smorodintsev 1970 | 15/1313 | 0/512 | | 3% | 12.24[0.73,204.86] |
| Wendel 1966 | 2/439 | 5/355 | + | 6.88% | 0.32[0.06,1.66] |
| Subtotal (95% CI) | 3001 | 2001 | • | 100% | 2.54[1.5,4.31] |
| Total events: 224 (Amantadine), 9 | 4 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.35; Chi ² =2 | 1.61, df=8(P=0.01); l ² =62. | 98% | | | |
| Test for overall effect: Z=3.46(P=0 |) | | | | |
| 1.3.3 Decreased CNS activity | | | | | |
| Callmander 1968 | 8/48 | 2/48 | | 10.33% | 4.6[0.92,22.93] |
| Millet 1982 | 2/10 | 0/5 | | 3.86% | 3.24[0.13,80.99] |
| Monto 1979 | | | | 20.45% | 1.42[0.83,2.41] |
| Oker-Blom 1970 | 42/144 | 32/142 | | 4.61% | |
| | 6/141 | 0/152 | | • | 14.63[0.82,262.14] |
| Plesnik 1977 | 82/387 | 30/419 | | 21.23% | 3.49[2.24,5.44] |
| Reuman 1989 | 37/317 | 24/159 | | 20.26% | 0.74[0.43,1.29] |
| Smorodintsev 1970 | 27/1313 | 14/512 | | 19.27% | 0.75[0.39,1.44] |
| Subtotal (95% CI) | 2360 | 1437 | | 100% | 1.73[0.86,3.45] |
| Total events: 204 (Amantadine), 1 | | | | | |
| Heterogeneity: Tau ² =0.54; Chi ² =2 Test for overall effect: Z=1.55(P=0 | | 79.25% | | | |
| | .12) | | | | |
| 1.3.4 Dermatological changes | | | | | |
| Callmander 1968 | 2/48 | 2/48 | _ | 48.09% | 1[0.14,7.4] |
| Monto 1979 | 2/144 | 1/142 | | 33.14% | 1.99[0.18,22.15] |
| Payler 1984 | 1/267 | 0/269 | | 18.77% | 3.03[0.12,74.81] |
| Subtotal (95% CI) | 459 | 459 | | 100% | 1.55[0.39,6.2] |
| Total events: 5 (Amantadine), 3 (F | | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.39 | | | | | |
| Test for overall effect: Z=0.61(P=0 | | | | | |
| | | | | | |
| | | | | | |

Analysis 1.3. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 3 Adverse effects.

Amantadine and rimantadine for influenza A in adults (Review)



| Study or subgroup | Amantadine | Placebo | Odds Ratio | Weight | Odds Ratio |
|---|---------------------------------------|--------------------|---------------------|------------------------------|---------------------|
| •••••) ••••• <u>•</u> ••• | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% Cl |
| Hayden 1981 | 41/67 | 13/66 | -+ | 14.37% | 6.43[2.95,14.03] |
| Pettersson 1980 | 20/246 | 34/255 | -+- | 16.56% | 0.58[0.32,1.03] |
| Plesnik 1977 | 95/574 | 38/559 | -+- | 18.45% | 2.72[1.83,4.04] |
| Quarles 1981 | 18/107 | 12/99 | _ + • | 14.29% | 1.47[0.67,3.22] |
| Reuman 1989 | 118/317 | 49/159 | | 18.36% | 1.33[0.89,2] |
| Smorodintsev 1970 | 94/1313 | 26/512 | +− | 17.97% | 1.44[0.92,2.25] |
| Subtotal (95% CI) | 2624 | 1650 | ◆ | 100% | 1.7[0.99,2.93] |
| Total events: 386 (Amantadine), 17 | 2 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.37; Chi ² =31. | 85, df=5(P<0.0001); l ² =8 | 34.3% | | | |
| Test for overall effect: Z=1.93(P=0.0 | 95) | | | | |
| | | | | | |
| 1.3.6 Withdrawals due to adverse | effects | | | | |
| Dolin 1982 | 23/145 | 7/148 | | 28.05% | 3.8[1.57,9.16] |
| Hayden 1981 | 1/67 | 1/66 | | 2.79% | 0.98[0.06,16.08] |
| Monto 1979 | 12/144 | 3/142 | - | 13.11% | 4.21[1.16,15.26] |
| Pettersson 1980 | 18/117 | 7/108 | | 25.9% | 2.62[1.05,6.56] |
| Plesnik 1977 | 8/574 | 5/559 | | 17.21% | 1.57[0.51,4.82] |
| Quarles 1981 | 6/107 | 4/99 | + | 12.94% | 1.41[0.39,5.16] |
| Subtotal (95% CI) | 1154 | 1122 | • | 100% | 2.54[1.6,4.06] |
| Total events: 68 (Amantadine), 27 (| Placebo) | | | | |
| Heterogeneity: Tau ² =0; Chi ² =3.35, o | df=5(P=0.65); I ² =0% | | | | |
| Test for overall effect: Z=3.93(P<0.0 | 0001) | | | | |
| | Fav | ours amantadine 0. | 01 0.1 1 10 10 | ⁰ Favours placebo | |

Analysis 1.4. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 4 Viral nasal shedding or persistence in upper airways at 2 to 5 days.

| Study or subgroup | Amantadine | Control | Risk Ratio | Weight | Risk Ratio |
|---------------------------------|------------------|---------|---------------------|--------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% Cl |
| Reuman 1989 | 36/59 | 18/20 | | 100% | 0.68[0.53,0.87] |
| Total (95% CI) | 59 | 20 | • | 100% | 0.68[0.53,0.87] |
| Total events: 36 (Amantadine | e), 18 (Control) | | | | |
| Heterogeneity: Not applicable | e | | | | |
| Test for overall effect: Z=3.04 | (P=0) | | | | |
| | _ | | 0.2 0.5 1 2 5 1 | 0 - | |

Favours amantadine 0.1 0.2 0.5 1 2 5 10 Favours control

Analysis 1.5. Comparison 1 Oral amantadine versus placebo (prophylaxis), Outcome 5 Influenza cases (asymptomatic).

| Study or subgroup | Amantadine | Placebo | | Risk Ratio | | | | | Weight | Risk Ratio | |
|-------------------|------------|-----------------|-----|------------|---------|------|---------|---|--------|-------------------|---------------------|
| | n/N | n/N | | | M-H, Ra | ndom | , 95% C | I | | | M-H, Random, 95% CI |
| Brady 1990 | 7/112 | 20/110 | | | | - | | | | 28.14% | 0.34[0.15,0.78] |
| Dolin 1982 | 5/113 | 5/132 | | | | + | | | | 20.13% | 1.17[0.35,3.93] |
| Monto 1979 | 25/142 | 32/139 | | | | - | | | | 36.29% | 0.76[0.48,1.22] |
| Nafta 1970 | 8/112 | 2/103 | | | | - | | + | - | 15.44% | 3.68[0.8,16.92] |
| | | | | | | | | | 10 | | |
| | Favo | ours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours placebo | |

Amantadine and rimantadine for influenza A in adults (Review)



| Study or subgroup | Amantadine | Placebo | | | Ri | sk Rat | io | | | Weight | Risk Ratio |
|---|--|-----------------|-----|---------------------|-----|--------|----|---|---------------------|-----------------|---------------|
| | n/N | n/N | | M-H, Random, 95% Cl | | | | | M-H, Random, 95% Cl | | |
| Total (95% CI) | 479 | 484 | | | | | • | | | 100% | 0.85[0.4,1.8] |
| Total events: 45 (Amantadine | e), 59 (Placebo) | | | | | | | | | | |
| Heterogeneity: Tau ² =0.35; Ch | i ² =8.18, df=3(P=0.04); l ² =63.3 | 31% | | | | | | | | | |
| Test for overall effect: Z=0.43 | (P=0.67) | | | | | | | | | | |
| | Fav | ours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours placebo | |

Comparison 2. Oral rimantadine versus placebo (prophylaxis)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|--|-------------------|-----------------------------|-------------------------------------|--------------------|
| 1 Influenza cases | 3 | 688 | Risk Ratio (M-H, Random, 95% CI) | 0.28 [0.08, 1.08] |
| 1.1 Unvaccinated population | 3 | 688 | Risk Ratio (M-H, Random, 95% CI) | 0.28 [0.08, 1.08] |
| 1.2 Vaccinated population | 0 | 0 | Risk Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 2 ILI cases | 3 | 688 | Risk Ratio (M-H, Random, 95% CI) | 0.65 [0.35, 1.20] |
| 2.1 Unvaccinated population | 3 | 688 | Risk Ratio (M-H, Random, 95% CI) | 0.65 [0.35, 1.20] |
| 2.2 Vaccinated population | 0 | 0 | Risk Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3 Adverse effects | 5 | | Odds Ratio (M-H, Random, 95% CI) | Subtotals only |
| 3.1 Gastrointestinal | 2 | 357 | Odds Ratio (M-H, Random, 95% CI) | 4.39 [1.43, 13.52] |
| 3.2 Increased CNS activity | 3 | 652 | Odds Ratio (M-H, Random, 95% CI) | 1.58 [0.78, 3.19] |
| 3.3 Decreased CNS activity | 2 | 243 | Odds Ratio (M-H, Random, 95% CI) | 1.31 [0.23, 7.50] |
| 3.4 Dermatological changes | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.5 All adverse effects | 3 | 558 | Odds Ratio (M-H, Random, 95% CI) | 1.96 [1.19, 3.22] |
| 3.6 Withdrawals due to adverse effects | 3 | 625 | Odds Ratio (M-H, Random, 95% CI) | 1.10 [0.48, 2.51] |
| 4 Influenza cases (asymptomatic) | 1 | 265 | Risk Ratio (M-H, Random, 95% CI) | 1.39 [0.45, 4.27] |

| Study or subgroup | Rimantadine | Placebo | Risk Ratio | Weight | Risk Ratio | |
|--|------------------------------------|------------------|---------------------|-----------------|---------------------|--|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% Cl | |
| 2.1.1 Unvaccinated population | | | | | | |
| Brady 1990 | 1/112 | 7/110 | + | 21.74% | 0.14[0.02,1.12] | |
| Dolin 1982 | 4/133 | 27/132 | — — | 36.25% | 0.15[0.05,0.41] | |
| Quarles 1981 | 15/102 | 20/99 | | 42.01% | 0.73[0.4,1.34] | |
| Subtotal (95% CI) | 347 | 341 | | 100% | 0.28[0.08,1.08] | |
| Total events: 20 (Rimantadine), 54 (Pl | acebo) | | | | | |
| Heterogeneity: Tau ² =1.01; Chi ² =8.82, o | df=2(P=0.01); I ² =77.3 | 1% | | | | |
| Test for overall effect: Z=1.84(P=0.07) | | | | | | |
| 2.1.2 Vaccinated population | | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable | |
| Total events: 0 (Rimantadine), 0 (Place | ebo) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Not applicable | | | | | | |
| Total (95% CI) | 347 | 341 | | 100% | 0.28[0.08,1.08] | |
| Total events: 20 (Rimantadine), 54 (Pl | acebo) | | | | | |
| Heterogeneity: Tau ² =1.01; Chi ² =8.82, o | df=2(P=0.01); I ² =77.3 | 1% | | | | |
| Test for overall effect: Z=1.84(P=0.07) | | | | | | |
| Test for subgroup differences: Not app | olicable | | | | | |
| | Fave | ours rimantadine | 0.02 0.1 1 10 50 | Favours placebo | | |

Analysis 2.1. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 1 Influenza cases.

Analysis 2.2. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 2 ILI cases.

| Study or subgroup | Rimantadine | Placebo | Risk Ratio | Weight | Risk Ratio |
|--|-------------------------------------|---------------------|---------------------|-------------------------------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI |
| 2.2.1 Unvaccinated population | | | | | |
| Brady 1990 | 19/112 | 21/110 | | 30.4% | 0.89[0.51,1.56] |
| Dolin 1982 | 19/133 | 54/132 | | 33.07% | 0.35[0.22,0.56] |
| Quarles 1981 | 40/102 | 44/99 | | 36.53% | 0.88[0.64,1.22] |
| Subtotal (95% CI) | 347 | 341 | • | 100% | 0.65[0.35,1.2] |
| Total events: 78 (Rimantadine), 119 (F | Placebo) | | | | |
| Heterogeneity: Tau ² =0.24; Chi ² =11.52 | , df=2(P=0); l ² =82.65% | | | | |
| Test for overall effect: Z=1.37(P=0.17) | | | | | |
| 2.2.2 Vaccinated population | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Rimantadine), 0 (Plac | ebo) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | | | | | |
| Total (95% CI) | 347 | 341 | • | 100% | 0.65[0.35,1.2] |
| Total events: 78 (Rimantadine), 119 (F | Placebo) | | | | |
| Heterogeneity: Tau ² =0.24; Chi ² =11.52 | , df=2(P=0); I ² =82.65% | | | | |
| Test for overall effect: Z=1.37(P=0.17) | | | | | |
| Test for subgroup differences: Not ap | plicable | | | | |
| | Favo | urs rimantadine 0.0 | 01 0.1 1 10 10 | ⁰⁰ Favours placebo | |

Analysis 2.3. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 3 Adverse effects.

| Study or subgroup | Rimantadine | Placebo | Odds Ratio | Weight | Odds Ratio |
|---|----------------------------------|---------|---------------------|--------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI |
| 2.3.1 Gastrointestinal | - / | | | | |
| Brady 1990 | 6/114 | 2/114 | | 48.09% | 3.11[0.61,15.75] |
| Hayden 1981 | 10/63 | 2/66 | | 51.91% | 6.04[1.27,28.77] |
| Subtotal (95% CI) | 177 | 180 | | 100% | 4.39[1.43,13.52] |
| Total events: 16 (Rimantadine), 4 | | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.33, | | | | | |
| Test for overall effect: Z=2.58(P=0. | .01) | | | | |
| 2.3.2 Increased CNS activity | | | | | |
| Brady 1990 | 4/114 | 2/114 | | 16.72% | 2.04[0.37,11.35] |
| Dolin 1982 | 9/147 | 6/148 | | 43.96% | 1.54[0.54,4.45] |
| Hayden 1981 | 8/63 | 6/66 | | 39.32% | 1.45[0.47,4.46] |
| Subtotal (95% CI) | 324 | 328 | ◆ | 100% | 1.58[0.78,3.19] |
| Total events: 21 (Rimantadine), 14 | 4 (Placebo) | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.11, | df=2(P=0.95); I ² =0% | | | | |
| Test for overall effect: Z=1.28(P=0. | .2) | | | | |
| 2.3.3 Decreased CNS activity | | | | | |
| Brady 1990 | 1/114 | 1/114 | | 39.36% | 1[0.06,16.18] |
| Millet 1982 | 7/10 | 3/5 | | 60.64% | 1.56[0.17,14.65] |
| Subtotal (95% CI) | 124 | 119 | | 100% | 1.31[0.23,7.5] |
| Total events: 8 (Rimantadine), 4 (I | | | | | [,] |
| Heterogeneity: Tau ² =0; Chi ² =0.06, | | | | | |
| Test for overall effect: Z=0.3(P=0.7 | | | | | |
| | | | | | |
| 2.3.4 Dermatological changes | 0 | 0 | | | Not estimable |
| Subtotal (95% CI) | | U | | | Notestimable |
| Total events: 0 (Rimantadine), 0 (I | Placebo) | | | | |
| Heterogeneity: Not applicable | bla | | | | |
| Test for overall effect: Not applica | ble | | | | |
| 2.3.5 All adverse effects | | | | | |
| Brady 1990 | 10/114 | 5/114 | + | 20.11% | 2.1[0.69,6.34] |
| Hayden 1981 | 18/63 | 13/66 | + | 36.94% | 1.63[0.72,3.69] |
| Quarles 1981 | 24/102 | 12/99 | | 42.94% | 2.23[1.05,4.76] |
| Subtotal (95% CI) | 279 | 279 | • | 100% | 1.96[1.19,3.22] |
| Total events: 52 (Rimantadine), 30 | | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.32, | | | | | |
| Test for overall effect: Z=2.66(P=0. | .01) | | | | |
| 2.3.6 Withdrawals due to advers | se effects | | | | |
| Dolin 1982 | 10/147 | 7/148 | | 68.47% | 1.47[0.54,3.97] |
| Hayden 1981 | 1/63 | 1/66 | _ | 8.67% | 1.05[0.06,17.13] |
| Quarles 1981 | 2/102 | 4/99 | | 22.86% | 0.48[0.09,2.65] |
| Subtotal (95% CI) | 312 | 313 | + | 100% | 1.1[0.48,2.51] |
| Total events: 13 (Rimantadine), 12 | 2 (Placebo) | | | | |
| Heterogeneity: Tau ² =0; Chi ² =1.24, | df=2(P=0.54); I ² =0% | | | | |
| Test for overall effect: Z=0.23(P=0. | 82) | | | | |

Amantadine and rimantadine for influenza A in adults (Review)

Analysis 2.4. Comparison 2 Oral rimantadine versus placebo (prophylaxis), Outcome 4 Influenza cases (asymptomatic).

| Study or subgroup | Rimantadine | Placebo | | Risk Ratio | | | Weight | Risk Ratio | | | |
|---|-------------|------------------|-----|------------|----------|------|----------|-------------------|----|-----------------|---------------------|
| | n/N | n/N | | | M-H, Rai | ndom | , 95% CI | | | | M-H, Random, 95% CI |
| Dolin 1982 | 7/133 | 5/132 | | | | | | _ | | 100% | 1.39[0.45,4.27] |
| Total (95% CI) | 133 | 132 | | | | | | - | | 100% | 1.39[0.45,4.27] |
| Total events: 7 (Rimantadine), 5 (Pla | cebo) | | | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=0.57(P=0.57) |) | | | | | | | | | | |
| | Favo | ours rimantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours placebo | |

Comparison 3. Oral amantadine versus oral rimantadine (prophylaxis)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|--|-------------------|-----------------------------|----------------------------------|-------------------|
| 1 Influenza cases | 2 | 455 | Risk Ratio (M-H, Random, 95% CI) | 0.89 [0.48, 1.65] |
| 1.1 Unvaccinated populations | 2 | 455 | Risk Ratio (M-H, Random, 95% CI) | 0.89 [0.48, 1.65] |
| 1.2 Vaccinated populations | 0 | 0 | Risk Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 2 ILI cases | 2 | 455 | Risk Ratio (M-H, Random, 95% CI) | 0.88 [0.57, 1.35] |
| 2.1 Unvaccinated populations | 2 | 455 | Risk Ratio (M-H, Random, 95% CI) | 0.88 [0.57, 1.35] |
| 2.2 Vaccinated populations | 0 | 0 | Risk Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3 Adverse effects | 3 | | Odds Ratio (M-H, Random, 95% CI) | Subtotals only |
| 3.1 Gastrointestinal | 1 | 130 | Odds Ratio (M-H, Random, 95% CI) | 1.28 [0.51, 3.16] |
| 3.2 Increased CNS activity | 2 | 422 | Odds Ratio (M-H, Random, 95% CI) | 3.11 [1.67, 5.78] |
| 3.3 Decreased CNS activity | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.4 Dermatological changes | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.5 All adverse effects | 2 | 339 | Odds Ratio (M-H, Random, 95% CI) | 1.60 [0.28, 9.26] |
| 3.6 Withdrawals due to adverse effects | 3 | 631 | Odds Ratio (M-H, Random, 95% CI) | 2.49 [1.26, 4.93] |

| Study or subgroup | Amantadine | Rimantadine | Risk Ratio | Weight | Risk Ratio |
|--|-------------------------------|------------------|----------------------|--------------------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI |
| 3.1.1 Unvaccinated populations | | | | | |
| Dolin 1982 | 2/113 | 4/133 | | 13.47% | 0.59[0.11,3.15] |
| Quarles 1981 | 15/107 | 15/102 | — <mark></mark> | 86.53% | 0.95[0.49,1.85] |
| Subtotal (95% CI) | 220 | 235 | | 100% | 0.89[0.48,1.65] |
| Total events: 17 (Amantadine), 19 (R | imantadine) | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.28, df | =1(P=0.6); I ² =0% | | | | |
| Test for overall effect: Z=0.36(P=0.72 |) | | | | |
| 3.1.2 Vaccinated populations | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Rim | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | 5 | | | | |
| Total (95% CI) | 220 | 235 | - | 100% | 0.89[0.48,1.65] |
| Total events: 17 (Amantadine), 19 (R | imantadine) | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.28, df | =1(P=0.6); l ² =0% | | | | |
| Test for overall effect: Z=0.36(P=0.72 |) | | | | |
| Test for subgroup differences: Not ap | oplicable | | | | |
| | Fa | vours amantadine | 0.1 0.2 0.5 1 2 5 10 | Favours rimantadin | e |

Analysis 3.1. Comparison 3 Oral amantadine versus oral rimantadine (prophylaxis), Outcome 1 Influenza cases.

Analysis 3.2. Comparison 3 Oral amantadine versus oral rimantadine (prophylaxis), Outcome 2 ILI cases.

| Study or subgroup | Amantadine | Rimantadine | Risk Ratio | Weight | Risk Ratio |
|--|-----------------------------------|------------------|--------------------|------------------------------------|---------------------|
| | n/N | n/N | M-H, Random, 95% C | I | M-H, Random, 95% CI |
| 3.2.1 Unvaccinated populations | | | | | |
| Dolin 1982 | 10/113 | 19/133 | | 27.66% | 0.62[0.3,1.28] |
| Quarles 1981 | 42/107 | 40/102 | | 72.34% | 1[0.71,1.4] |
| Subtotal (95% CI) | 220 | 235 | - | 100% | 0.88[0.57,1.35] |
| Total events: 52 (Amantadine), 59 (Ri | mantadine) | | | | |
| Heterogeneity: Tau ² =0.04; Chi ² =1.44, | df=1(P=0.23); I ² =30. | 47% | | | |
| Test for overall effect: Z=0.6(P=0.55) | | | | | |
| 3.2.2 Vaccinated populations | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Rima | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | | | | | |
| Total (95% CI) | 220 | 235 | • | 100% | 0.88[0.57,1.35] |
| Total events: 52 (Amantadine), 59 (Ri | mantadine) | | | | |
| Heterogeneity: Tau ² =0.04; Chi ² =1.44, | df=1(P=0.23); I ² =30. | 47% | | | |
| Test for overall effect: Z=0.6(P=0.55) | | | | | |
| Test for subgroup differences: Not ap | plicable | | | | |
| | Fa | vours amantadine | 0.1 0.2 0.5 1 2 | ^{5 10} Favours rimantadir | ie |

| Study or subgroup | Amantadine | Rimantadine | Odds Ratio | Weight | Odds Ratio | |
|--|-------------------------------------|-------------|---------------------|--------|---------------------|--|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% CI | |
| 3.3.1 Gastrointestinal | | | | | | |
| Hayden 1981 | 13/67 | 10/63 | | 100% | 1.28[0.51,3.16] | |
| Subtotal (95% CI) | 67 | 63 | | 100% | 1.28[0.51,3.16] | |
| Total events: 13 (Amantadine), 10 (R | Rimantadine) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Z=0.53(P=0.6) | | | | | | |
| 3.3.2 Increased CNS activity | | | | | | |
| Dolin 1982 | 19/145 | 9/147 | | 53.37% | 2.31[1.01,5.3] | |
| Hayden 1981 | 26/67 | 8/63 | │ | 46.63% | 4.36[1.79,10.61] | |
| Subtotal (95% CI) | 212 | 210 | | 100% | 3.11[1.67,5.78] | |
| Total events: 45 (Amantadine), 17 (R | Rimantadine) | | | | | |
| Heterogeneity: Tau ² =0.01; Chi ² =1.05 | , df=1(P=0.31); l ² =4.3 | 2% | | | | |
| Test for overall effect: Z=3.58(P=0) | | | | | | |
| 3.3.3 Decreased CNS activity | | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable | |
| Total events: 0 (Amantadine), 0 (Rim | nantadine) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Not applicable | e | | | | | |
| 3.3.4 Dermatological changes | | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable | |
| Total events: 0 (Amantadine), 0 (Rim | nantadine) | | | | | |
| Heterogeneity: Not applicable | · | | | | | |
| Test for overall effect: Not applicable | e | | | | | |
| 3.3.5 All adverse effects | | | | | | |
| Hayden 1981 | 41/67 | 18/63 | _ | 49.7% | 3.94[1.89,8.22] | |
| Quarles 1981 | 18/107 | 24/102 | | 50.3% | 0.66[0.33,1.3] | |
| Subtotal (95% CI) | 174 | 165 | _ | 100% | 1.6[0.28,9.26] | |
| Total events: 59 (Amantadine), 42 (R | | | | | | |
| Heterogeneity: Tau ² =1.47; Chi ² =12.2 | | 4% | | | | |
| Test for overall effect: Z=0.53(P=0.6) | | | | | | |
| 3.3.6 Withdrawals due to adverse | effects | | | | | |
| Dolin 1982 | 23/145 | 10/147 | — — | 76.35% | 2.58[1.18,5.64] | |
| Hayden 1981 | 1/67 | 1/63 | | 5.98% | 0.94[0.06,15.35] | |
| Quarles 1981 | 6/107 | 2/102 | | 17.68% | 2.97[0.59,15.07] | |
| Subtotal (95% CI) | 319 | 312 | | 100% | 2.49[1.26,4.93] | |
| Total events: 30 (Amantadine), 13 (R | | | | 20070 | | |
| Heterogeneity: Tau ² =0; Chi ² =0.52, df | | | | | | |
| Test for overall effect: Z=2.62(P=0.01 | | | | | | |

Analysis 3.3. Comparison 3 Oral amantadine versus oral rimantadine (prophylaxis), Outcome 3 Adverse effects.

Comparison 4. Oral amantadine versus placebo (treatment)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|---|-------------------|-----------------------------|--------------------------------------|----------------------|
| 1 Duration of fever (37 degrees centigrade or more) in days | 10 | 542 | Mean Difference (IV, Random, 95% CI) | -0.99 [-1.26, -0.71] |
| 2 Cases with fever at 48 hours | 2 | 85 | Risk Ratio (M-H, Random, 95% CI) | 0.21 [0.07, 0.66] |
| 3 Adverse effects | 4 | | Odds Ratio (M-H, Random, 95% CI) | Subtotals only |
| 3.1 Gastrointestinal | 3 | 494 | Odds Ratio (M-H, Random, 95% CI) | 1.34 [0.32, 5.61] |
| 3.2 Increased CNS activity | 2 | 465 | Odds Ratio (M-H, Random, 95% CI) | 0.77 [0.23, 2.53] |
| 3.3 Decreased CNS activity | 3 | 491 | Odds Ratio (M-H, Random, 95% CI) | 0.65 [0.31, 1.38] |
| 3.4 Dermatological changes | 2 | 465 | Odds Ratio (M-H, Random, 95% CI) | 1.40 [0.14, 13.78] |
| 3.5 Withdrawals due to ad- verse effects | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 4 Duration of hospital stay (in days) | 1 | 36 | Mean Difference (IV, Random, 95% CI) | -0.90 [-2.20, 0.40] |
| 5 Viral nasal shedding or per- sistence in upper airways at 2 to 5 days | 3 | 170 | Risk Ratio (M-H, Random, 95% CI) | 0.97 [0.76, 1.24] |

Analysis 4.1. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 1 Duration of fever (37 degrees centigrade or more) in days.

| Study or subgroup | Am | antadine | Р | lacebo | Mean Difference | Weight | Mean Difference |
|--|------------------|------------------|--------|------------------|-----------------|---------------------------|--------------------|
| | Ν | Mean(SD) | Ν | Mean(SD) | Random, 95% CI | | Random, 95% CI |
| Galbraith 1971 | 72 | 1.9 (2.5) | 81 | 3.1 (2.5) | -+- | 11.82% | -1.18[-1.97,-0.39] |
| Hornick 1969a | 12 | 2.8 (1.5) | 15 | 3.8 (1.5) | -+ | 6.06% | -1.08[-2.19,0.03] |
| Hornick 1969b | 21 | 2.5 (1.4) | 28 | 3.3 (1.4) | -+- | 12.21% | -0.81[-1.59,-0.03] |
| Hornick 1969c | 23 | 2.7 (1.2) | 20 | 3.7 (1.2) | -+- | 15.14% | -0.97[-1.67,-0.27] |
| Hornick 1969d | 17 | 2.1 (1.6) | 17 | 3.4 (1.6) | -+ | 6.2% | -1.34[-2.44,-0.24] |
| Kitamoto 1968 | 14 | 2 (2.8) | 23 | 2.7 (1.4) | —+ <u> </u> | 3.09% | -0.74[-2.29,0.81] |
| Kitamoto 1971 | 38 | 1.5 (1.2) | 46 | 2.6 (2) | - | 15.83% | -1.1[-1.79,-0.41] |
| Knight 1970 | 13 | 1.9 (1.5) | 16 | 3 (1.5) | -+- | 6.52% | -1.12[-2.19,-0.05] |
| Máté 1970 | 20 | 3.2 (1.4) | 16 | 3.9 (1.6) | -+- | 7.51% | -0.7[-1.7,0.3] |
| Wingfield 1969 | 20 | 1.1 (1.2) | 30 | 1.9 (1.2) | -+- | 15.62% | -0.83[-1.52,-0.14] |
| Total *** | 250 | | 292 | | • | 100% | -0.99[-1.26,-0.71] |
| Heterogeneity: Tau ² =0; Chi ² = | 1.63, df=9(P=1); | ² =0% | | | | | |
| Test for overall effect: Z=7.08 | (P<0.0001) | | | | | | |
| | | | Favour | s amantadine -10 | -5 0 5 | ¹⁰ Favours pla | cebo |

Analysis 4.2. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 2 Cases with fever at 48 hours.

| Study or subgroup | Amantadine | Placebo | | Risk Ratio | | | Weight | Risk Ratio | | | |
|---|---------------------------------------|-----------------|-----|------------|---------|------|----------|-------------------|----|-----------------|---------------------|
| | n/N | n/N | | | M-H, Ra | ndom | , 95% CI | | | | M-H, Random, 95% Cl |
| van Voris 1981 | 1/14 | 6/12 | • | | | _ | | | | 32.67% | 0.14[0.02,1.03] |
| Wingfield 1969 | 2/20 | 15/39 | ← | - | | | | | | 67.33% | 0.26[0.07,1.03] |
| Total (95% CI) | 34 | 51 | | | | | | | | 100% | 0.21[0.07,0.66] |
| Total events: 3 (Amantadine), | 21 (Placebo) | | | | | | | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0 | .24, df=1(P=0.63); I ² =0% | | | | | | | | | | |
| Test for overall effect: Z=2.68(H | P=0.01) | | | | | | | | | | |
| | Favo | ours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours placebo | |

Analysis 4.3. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 3 Adverse effects.

| Study or subgroup | Amantadine | Placebo | Odds Ratio | Weight | Odds Ratio |
|---|---|-----------------------------|---------------------|-------------------------------|---------------------|
| | n/N | n/N | M-H, Random, 95% CI | | M-H, Random, 95% CI |
| 4.3.1 Gastrointestinal | | | | | |
| Kitamoto 1968 | 16/107 | 5/89 | | 45.96% | 2.95[1.04,8.42] |
| Kitamoto 1971 | 17/120 | 29/149 | | 54.04% | 0.68[0.36,1.31] |
| Knight 1970 | 0/13 | 0/16 | | | Not estimable |
| Subtotal (95% CI) | 240 | 254 | | 100% | 1.34[0.32,5.61] |
| Total events: 33 (Amantadine), | 34 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.88; Chi ² = | =5.43, df=1(P=0.02); l ² =81.589 | 6 | | | |
| Test for overall effect: Z=0.4(P=0 | 0.69) | | | | |
| 4.3.2 Increased CNS activity | | | | | |
| Kitamoto 1968 | 4/107 | 2/89 | | - 37.23% | 1.69[0.3,9.45] |
| Kitamoto 1971 | 4/120 | 10/149 | _ | 62.77% | 0.48[0.15,1.57] |
| Subtotal (95% CI) | 227 | 238 | | 100% | 0.77[0.23,2.53] |
| Total events: 8 (Amantadine), 1 | 2 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.22; Chi ² = | =1.4, df=1(P=0.24); l ² =28.35% | | | | |
| Test for overall effect: Z=0.44(P= | =0.66) | | | | |
| 4.3.3 Decreased CNS activity | | | | | |
| Kitamoto 1968 | 77/107 | 76/89 | _ | 40.26% | 0.44[0.21,0.91] |
| Kitamoto 1971 | 54/120 | 86/149 | | 50.73% | 0.6[0.37,0.97] |
| van Voris 1981 | 5/14 | 1/12 | | 9.01% | 6.11[0.6,62.23] |
| Subtotal (95% CI) | 241 | 250 | | 100% | 0.65[0.31,1.38] |
| Total events: 136 (Amantadine) | , 163 (Placebo) | | | | |
| Heterogeneity: Tau ² =0.23; Chi ² = | =4.56, df=2(P=0.1); l ² =56.13% | | | | |
| Test for overall effect: Z=1.12(P= | =0.26) | | | | |
| 4.3.4 Dermatological changes | i | | | | |
| Kitamoto 1968 | 2/107 | 0/89 | | 52.39% | 4.24[0.2,89.51] |
| Kitamoto 1971 | 0/120 | 1/149 | | 47.61% | 0.41[0.02,10.18] |
| Subtotal (95% CI) | 227 | 238 | | 100% | 1.4[0.14,13.78] |
| Total events: 2 (Amantadine), 1 | (Placebo) | | | | |
| Heterogeneity: Tau ² =0.18; Chi ² = | =1.07, df=1(P=0.3); l ² =6.73% | | | | |
| Test for overall effect: Z=0.29(P= | =0.78) | | | | |
| 4.3.5 Withdrawals due to adve | erse effects | | | | |
| | Envio | urs amantadine ⁰ | 1 0.2 0.5 1 2 5 1 | ¹⁰ Favours placebo | |

Amantadine and rimantadine for influenza A in adults (Review)



| Study or subgroup | Amantadine | Placebo | | | Od | lds Ra | tio | | | Weight | Odds Ratio |
|---|------------|------------------|-----|-----|---------|--------|-----------|---|----|-----------------|---------------------|
| | n/N | n/N | | | M-H, Ra | ndom | i, 95% Cl | | | | M-H, Random, 95% CI |
| Subtotal (95% CI) | 0 | 0 | | | | | | | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Placeb | o) | | | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Not applicable | | | | | | | | | | | |
| | Fa | vours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours placebo | |

Analysis 4.4. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 4 Duration of hospital stay (in days).

| Study or subgroup | Am | antadine | Р | lacebo | | M | ean Differer | nce | | Weight | Mean Difference |
|---|----|-----------|--------|--------------|-----|----|--------------|-----|----|----------------|-----------------|
| | Ν | Mean(SD) | Ν | Mean(SD) | | Ra | ndom, 95% | CI | | | Random, 95% CI |
| Máté 1970 | 20 | 5.7 (1.8) | 16 | 6.6 (2.1) | | | | | | 100% | -0.9[-2.2,0.4] |
| Total *** | 20 | | 16 | | | | • | | | 100% | -0.9[-2.2,0.4] |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=1.36(P=0.17) | | | | | | | | | | | |
| | | | Favour | s amantadine | -10 | -5 | 0 | 5 | 10 | Favours placeb | 0 |

Analysis 4.5. Comparison 4 Oral amantadine versus placebo (treatment), Outcome 5 Viral nasal shedding or persistence in upper airways at 2 to 5 days.

| Study or subgroup | Amantadine | Amantadine Control | | | | sk Rat | io | | | Weight | Risk Ratio |
|---|--|--------------------|-----|-----|---------|--------|--------|---|----|-----------------|---------------------|
| | n/N | n/N | | | M-H, Ra | ndom, | 95% CI | | | | M-H, Random, 95% CI |
| Hornick 1969a | 36/38 | 33/36 | | | | - | | | | 46.13% | 1.03[0.91,1.17] |
| van Voris 1981 | 3/14 | 9/12 | ← | + | | - | | | | 4.94% | 0.29[0.1,0.82] |
| Wingfield 1969 | 23/23 | 45/47 | | | | + | | | | 48.93% | 1.03[0.95,1.13] |
| Total (95% CI) | 75 | 95 | | | | • | | | | 100% | 0.97[0.76,1.24] |
| Total events: 62 (Amantadine | e), 87 (Control) | | | | | | | | | | |
| Heterogeneity: Tau ² =0.03; Ch | i ² =12.22, df=2(P=0); l ² =83.64% | | | | | | | | | | |
| Test for overall effect: Z=0.25 | (P=0.81) | | | | | | | | | | |
| | Favou | rs amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours control | |

Comparison 5. Oral rimantadine versus placebo (treatment)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|---|-------------------|-----------------------------|--------------------------------------|----------------------|
| 1 Duration of fever (37 degrees centigrade or more) in days | 3 | 82 | Mean Difference (IV, Random, 95% CI) | -1.24 [-1.71, -0.76] |
| 2 Cases with fever at 48 hours | 4 | 122 | Risk Ratio (M-H, Random, 95% CI) | 0.16 [0.05, 0.53] |
| 3 Adverse effects | 2 | | Odds Ratio (M-H, Random, 95% CI) | Subtotals only |

Amantadine and rimantadine for influenza A in adults (Review)

Trusted evidence. Informed decisions. Better health.

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|---|-------------------|-----------------------------|----------------------------------|-------------------|
| 3.1 Gastrointestinal | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.2 Increased CNS activity | 1 | 14 | Odds Ratio (M-H, Random, 95% CI) | 1.0 [0.10, 10.17] |
| 3.3 Decreased CNS activity | 1 | 31 | Odds Ratio (M-H, Random, 95% CI) | 0.20 [0.01, 5.24] |
| 3.4 Dermatological changes | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.5 Withdrawals due to adverse effects | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 4 Viral nasal shedding or persis- tence in upper airways at 2 to 5 days | 3 | 152 | Risk Ratio (M-H, Random, 95% CI) | 0.68 [0.30, 1.53] |

Analysis 5.1. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 1 Duration of fever (37 degrees centigrade or more) in days.

| Study or subgroup | Rim | antadine | P | lacebo | | Mean Difference | | Weight | | Mean Difference |
|---|------------------|------------------------|---------|-------------|-----|-----------------|------------|--------|---------------|--------------------|
| | N | Mean(SD) | Ν | Mean(SD) | | Rand | om, 95% CI | | | Random, 95% Cl |
| Hayden 1986 | 7 | 1.3 (0.9) | 7 | 2.8 (0.3) | | -# | F | | 42.88% | -1.54[-2.26,-0.82] |
| Rabinovich 1969 | 9 | 1.2 (0.9) | 9 | 2.1 (2.1) | | | • | | 9.93% | -0.9[-2.4,0.6] |
| Wingfield 1969 | 20 | 0.9 (1.2) | 30 | 1.9 (1.2) | | 4 | ₽ | | 47.19% | -1.03[-1.72,-0.34] |
| Total *** | 36 | | 46 | | | | • | | 100% | -1.24[-1.71,-0.76] |
| Heterogeneity: Tau ² =0; Chi ² =1 | 1.21, df=2(P=0.5 | 5); I ² =0% | | | | | | | | |
| Test for overall effect: Z=5.11(| P<0.0001) | | | | | | | | | |
| | | | Favours | rimantadine | -10 | -5 | 0 5 | 10 | Favours place | bo |

Analysis 5.2. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 2 Cases with fever at 48 hours.

| Study or subgroup | Rimantadine | Placebo | | Ris | k Ratio | | Weight | Risk Ratio |
|---|---|------------------|-----|----------|-------------|----|-----------------|---------------------|
| | n/N | n/N | | M-H, Ran | dom, 95% Cl | | | M-H, Random, 95% Cl |
| Hayden 1986 | 0/7 | 7/7 | ← | | _ | | 16.22% | 0.07[0,0.98] |
| Rabinovich 1969 | 0/9 | 7/9 | ← | | _ | | 15.86% | 0.07[0,1.02] |
| van Voris 1981 | 0/19 | 6/12 | ← | | | | 15.24% | 0.05[0,0.81] |
| Wingfield 1969 | 3/20 | 15/39 | - | | + | | 52.67% | 0.39[0.13,1.19] |
| Total (95% CI) | 55 | 67 | | | | | 100% | 0.16[0.05,0.53] |
| Total events: 3 (Rimantadine), | 35 (Placebo) | | | | | | | |
| Heterogeneity: Tau ² =0.37; Chi ² | ² =3.89, df=3(P=0.27); l ² =22.78 | 3% | | | | | | |
| Test for overall effect: Z=3.01(F | P=0) | | | | | | | |
| | Favo | ours rimantadine | 0.1 | 0.2 0.5 | 1 2 5 | 10 | Favours placebo | |

| Study or subgroup | Rimantadine | Placebo | Odds Ratio | Weight | Odds Ratio | |
|--|-------------|---------|---------------------|--------|---------------------|--|
| | n/N | n/N | M-H, Random, 95% CI | | M-H, Random, 95% Cl | |
| 5.3.1 Gastrointestinal | | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable | |
| Total events: 0 (Rimantadine), 0 (Pla | acebo) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Not applicabl | e | | | | | |
| 5.3.2 Increased CNS activity | | | | | | |
| Hayden 1986 | 2/7 | 2/7 | ← | 100% | 1[0.1,10.17] | |
| Subtotal (95% CI) | 7 | 7 | | 100% | 1[0.1,10.17] | |
| Total events: 2 (Rimantadine), 2 (Pla | acebo) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Not applicabl | e | | | | | |
| 5.3.3 Decreased CNS activity | | | | | | |
| van Voris 1981 | 0/19 | 1/12 | | 100% | 0.2[0.01,5.24] | |
| Subtotal (95% CI) | 19 | 12 | | 100% | 0.2[0.01,5.24] | |
| Total events: 0 (Rimantadine), 1 (Pla | acebo) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Z=0.97(P=0.33 | 3) | | | | | |
| 5.3.4 Dermatological changes | | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable | |
| Total events: 0 (Rimantadine), 0 (Pla | acebo) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Not applicabl | e | | | | | |
| 5.3.5 Withdrawals due to adverse | effects | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable | |
| Total events: 0 (Rimantadine), 0 (Pla | acebo) | | | | | |
| Heterogeneity: Not applicable | | | | | | |
| Test for overall effect: Not applicabl | e | | | | | |

Analysis 5.3. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 3 Adverse effects.

Analysis 5.4. Comparison 5 Oral rimantadine versus placebo (treatment), Outcome 4 Viral nasal shedding or persistence in upper airways at 2 to 5 days.

| Study or subgroup | Rimantadine | Control | | Risk Ratio | | | | Weight | Risk Ratio | | |
|--|--|------------------|-----|------------|---------|------|----------|--------|-------------------|-----------------|---------------------|
| | n/N | n/N | | | M-H, Ra | ndom | , 95% CI | | | | M-H, Random, 95% Cl |
| Hayden 1986 | 16/26 | 23/24 | | | | - | | | | 34.97% | 0.64[0.47,0.88] |
| van Voris 1981 | 6/19 | 9/12 | | | - | -1 | | | | 28.27% | 0.42[0.2,0.88] |
| Wingfield 1969 | 24/24 | 45/47 | | | | # | | | | 36.76% | 1.03[0.95,1.13] |
| Total (95% CI) | 69 | 83 | | | | | - | | | 100% | 0.68[0.3,1.53] |
| Total events: 46 (Rimantadine | e), 77 (Control) | | | | | | | | | | |
| Heterogeneity: Tau ² =0.46; Chi | ² =41.39, df=2(P<0.0001); l ² =9 | 5.17% | | | | | | | | | |
| Test for overall effect: Z=0.94(I | P=0.35) | | | | 1 | | | | | | |
| | Favo | ours rimantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours control | |

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|---|-------------------|-----------------------------|---|----------------------|
| 1 Duration of fever (37 degrees centigrade or more) in days | 1 | 40 | Mean Difference (IV, Random, 95% CI) | 0.20 [-0.56, 0.96] |
| 2 Cases with fever at 48 hours | 2 | 73 | Risk Ratio (M-H, Random, 95% CI) | 0.99 [0.23, 4.37] |
| 3 Adverse effects | 1 | | Odds Ratio (M-H, Random, 95% CI) | Subtotals only |
| 3.1 Gastrointestinal | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.2 Increased CNS activity | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.3 Decreased CNS activity | 1 | 33 | Odds Ratio (M-H, Random, 95% CI) | 22.58 [1.13, 452.21] |
| 3.4 Dermatological changes | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 3.5 Withdrawals due to adverse effects | 0 | 0 | Odds Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |

Comparison 6. Oral amantadine versus oral rimantadine (treatment)

Analysis 6.1. Comparison 6 Oral amantadine versus oral rimantadine (treatment), Outcome 1 Duration of fever (37 degrees centigrade or more) in days.

| Study or subgroup | Am | Amantadine Rimantad | | antadine | tadine Mean Difference | | | | | Weight | Mean Difference |
|--|----|---------------------|---------|------------|------------------------|-----|-----------|----|----|-------------|-----------------|
| | Ν | Mean(SD) | Ν | Mean(SD) | | Rai | ndom, 95% | CI | | | Random, 95% CI |
| Wingfield 1969 | 20 | 1.1 (1.2) | 20 | 0.9 (1.2) | | | | | | 100% | 0.2[-0.56,0.96] |
| Total *** | 20 | | 20 | | | | • | | | 100% | 0.2[-0.56,0.96] |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=0.52(P=0.6) | | | | | | | | | | | |
| | | | Favours | amantadine | -10 | -5 | 0 | 5 | 10 | Favours rim | antadine |

Analysis 6.2. Comparison 6 Oral amantadine versus oral rimantadine (treatment), Outcome 2 Cases with fever at 48 hours.

| Study or subgroup | Amantadine | Rimantadine | | Risk Ratio | | | | | Weight | Risk Ratio |
|-------------------|------------|-----------------|-----|----------------|----------|-----------|---|---------------|---------------------|---------------------|
| | n/N | n/N | | M-H, Rai | ndom | n, 95% CI | | | | M-H, Random, 95% CI |
| van Voris 1981 | 1/14 | 0/19 | - | | | | • | \rightarrow | 22.34% | 4[0.17,91.48] |
| Wingfield 1969 | 2/20 | 3/20 | | <mark>.</mark> | \vdash | | - | | 77.66% | 0.67[0.12,3.57] |
| | Fav | ours amantadine | 0.1 | 0.2 0.5 | 1 | 2 | 5 | 10 | Favours rimantadine | |

Amantadine and rimantadine for influenza A in adults (Review)



| Study or subgroup | udy or subgroup Amantadine | | | Risk Ratio | | | | | | Weight | Risk Ratio | |
|---|--------------------------------------|------------------|-----|-------------|-----|---|-----------|---|----|---------------------|---------------------|--|
| | n/N | n/N | | M-H, Random | | | m, 95% Cl | | | | M-H, Random, 95% Cl | |
| Total (95% CI) | 34 | 39 | | _ | | | | - | | 100% | 0.99[0.23,4.37] | |
| Total events: 3 (Amantadine), 3 | (Rimantadine) | | | | | | | | | | | |
| Heterogeneity: Tau ² =0; Chi ² =0.9 | 99, df=1(P=0.32); I ² =0% | | | | | | | | | | | |
| Test for overall effect: Z=0.01(P= | =0.99) | | | | 1 | | | | | | | |
| | Fa | vours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours rimantadine | | |

Analysis 6.3. Comparison 6 Oral amantadine versus oral rimantadine (treatment), Outcome 3 Adverse effects.

| Study or subgroup | Amantadine | Rimantadine | Odds Ratio | Weight | Odds Ratio |
|---|------------|-------------|---------------------|--------|---------------------|
| | n/N | n/N | M-H, Random, 95% Cl | | M-H, Random, 95% Cl |
| 6.3.1 Gastrointestinal | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Rim | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | 5 | | | | |
| 6.3.2 Increased CNS activity | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Rim | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | 2 | | | | |
| 6.3.3 Decreased CNS activity | | | | | |
| van Voris 1981 | 5/14 | 0/19 | | 100% | 22.58[1.13,452.21] |
| Subtotal (95% CI) | 14 | 19 | | 100% | 22.58[1.13,452.21] |
| Total events: 5 (Amantadine), 0 (Rim | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Z=2.04(P=0.04 | .) | | | | |
| 6.3.4 Dermatological changes | | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Rim | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | 9 | | | | |
| 6.3.5 Withdrawals due to adverse | effects | | | | |
| Subtotal (95% CI) | 0 | 0 | | | Not estimable |
| Total events: 0 (Amantadine), 0 (Rim | antadine) | | | | |
| Heterogeneity: Not applicable | | | | | |
| Test for overall effect: Not applicable | e | | | | |



Comparison 7. Oral or inhaled amantadine versus placebo or aspirin

| Outcome or subgroup title | No. of studies | No. of par- ticipants | Statistical method | Effect size |
|--|-------------------|--------------------------|-------------------------------------|----------------------|
| 1 Viral nasal shedding or persistence in upper airways at 2 to 5 days | 5 | 237 | Risk Ratio (M-H, Random, 95% CI) | 0.94 [0.74, 1.19] |

Analysis 7.1. Comparison 7 Oral or inhaled amantadine versus placebo or aspirin, Outcome 1 Viral nasal shedding or persistence in upper airways at 2 to 5 days.

| Study or subgroup | Amantadine | Control | | R | isk Rat | io | | | Weight | Risk Ratio | |
|---|---|-----------------|-----|---------|---------|----------|---|----|-----------------|---------------------|--|
| | n/N | n/N | | M-H, R | andom | , 95% CI | | | | M-H, Random, 95% Cl | |
| Hayden 1980 | 5/9 | 4/11 | | - | | + | - | | 5.01% | 1.53[0.58,4.05] | |
| Hornick 1969a | 36/38 | 33/36 | | | + | | | | 36.68% | 1.03[0.91,1.17] | |
| van Voris 1981 | 3/14 | 9/12 | - | + | - | | | | 4.37% | 0.29[0.1,0.82] | |
| Wingfield 1969 | 23/23 | 45/47 | | | • | | | | 38.63% | 1.03[0.95,1.13] | |
| Younkin 1983 | 15/30 | 12/17 | | | • | | | | 15.31% | 0.71[0.44,1.13] | |
| Total (95% CI) | 114 | 123 | | | • | | | | 100% | 0.94[0.74,1.19] | |
| Total events: 82 (Amantadine) | , 103 (Control) | | | | | | | | | | |
| Heterogeneity: Tau ² =0.03; Chi ² | ² =16.81, df=4(P=0); I ² =76.2% | | | | | | | | | | |
| Test for overall effect: Z=0.52(F | P=0.6) | | | | | | | | | | |
| | Favo | ours amantadine | 0.1 | 0.2 0.5 | 1 | 2 | 5 | 10 | Favours control | | |

Comparison 8. Oral amantadine versus standard medication (treatment)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|--|-------------------|-----------------------------|---|--------------------|
| 1 Duration of fever (37 degrees centigrade or more) in days | 2 | 78 | Mean Difference (IV, Random, 95% CI) | 0.25 [-0.37, 0.87] |
| 2 Adverse effect - insomnia | 1 | 47 | Odds Ratio (M-H, Random, 95% CI) | 0.92 [0.26, 3.20] |
| 3 Viral nasal shedding or persistence in up- per airways at 2 to 5 days | 1 | 47 | Risk Ratio (M-H, Random, 95% CI) | 0.71 [0.44, 1.13] |

Analysis 8.1. Comparison 8 Oral amantadine versus standard medication (treatment), Outcome 1 Duration of fever (37 degrees centigrade or more) in days.

| Study or subgroup | Ama | antadine | Standard medication | | | Mean Difference | | | Mean Difference Weight | | Weight | Mean Difference |
|-------------------|-----|-----------|---------------------|--------------|-----|-----------------|-----------|----|------------------------|-------------|------------------|-----------------|
| | N | Mean(SD) | Ν | Mean(SD) | | Ra | ndom, 95% | CI | | | Random, 95% Cl | |
| Ito 2000 | 26 | 2.5 (1.5) | 23 | 2.7 (1.4) | | | - | | | 33.33% | -0.2[-1.01,0.61] | |
| Younkin 1983 | 20 | 0.9 (0.4) | 9 | 0.4 (0.4) | | | + | | | 66.67% | 0.47[0.18,0.76] | |
| | | | Favour | s amantadine | -10 | -5 | 0 | 5 | 10 | Favours sta | ndard med. | |

Amantadine and rimantadine for influenza A in adults (Review)



| Study or subgroup | Am | antadine | e Standard medication | | Mean Difference | | | | Weight | Mean Difference | |
|--|-----------------------------|--------------------------------|--------------------------|----------|-----------------|----|--------------|------|--------|------------------|--|
| | Ν | Mean(SD) | Ν | Mean(SD) | | Ra | ndom, 95% Cl | | | Random, 95% Cl | |
| Total *** | 46 | | 32 | | | | • | | 100% | 0.25[-0.37,0.87] | |
| Heterogeneity: Tau ² =0.13; Chi | ² =2.32, df=1(P= | =0.13); I ² =56.83% | | | | | | | | | |
| Test for overall effect: Z=0.78(| P=0.43) | | | | | | | | | | |
| | | | _ | | 10 | | 0 | - 10 | | | |

Favours amantadine ⁻¹⁰ ⁻⁵ ⁰ ⁵ ¹⁰ Favours standard med.

Analysis 8.2. Comparison 8 Oral amantadine versus standard medication (treatment), Outcome 2 Adverse effect - insomnia.

| Study or subgroup | Amantadine | Aspirin | pirin | | | ds Rat | tio | | | Weight | Odds Ratio |
|---------------------------------------|------------|-----------------|-------|-----|----------|--------|----------|---|----|-----------------|---------------------|
| | n/N | n/N | | | M-H, Rai | ndom | , 95% CI | | | | M-H, Random, 95% CI |
| Younkin 1983 | 10/30 | 6/17 | | _ | | | | | | 100% | 0.92[0.26,3.2] |
| Total (95% CI) | 30 | 17 | | _ | | | | | | 100% | 0.92[0.26,3.2] |
| Total events: 10 (Amantadine), 6 (A | Aspirin) | | | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=0.14(P=0.8 | 39) | | | | | | 1 | | | | |
| | Fav | ours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours aspirin | |

Analysis 8.3. Comparison 8 Oral amantadine versus standard medication (treatment), Outcome 3 Viral nasal shedding or persistence in upper airways at 2 to 5 days.

| Study or subgroup | Amantadine | Control | | Risk Ratio | | | | Weight | Risk Ratio | |
|-----------------------------------|--------------|-----------------|--------|------------|------|----------|---|--------|-------------------|---------------------|
| | n/N | n/N | | M-H, Ra | ndom | , 95% CI | | | | M-H, Random, 95% Cl |
| Younkin 1983 | 15/30 | 12/17 | | | + | | | | 100% | 0.71[0.44,1.13] |
| Total (95% CI) | 30 | 17 | | | | | | | 100% | 0.71[0.44,1.13] |
| Total events: 15 (Amantadine), | 12 (Control) | | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | | |
| Test for overall effect: Z=1.43(P | =0.15) | | | | | | | | | |
| | Favo | ours amantadine | 0.1 0. | 2 0.5 | 1 | 2 | 5 | 10 | Favours control | |

Comparison 9. Inhaled amantadine versus placebo (treatment)

| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|---|-------------------|-----------------------------|---|-------------------------|
| 1 Symptoms score in confirmed cases (respi- ratory illness) | 1 | 20 | Mean Difference (IV, Random, 95% CI) | -1.0 [-3.64, 1.64] |
| 2 Symptoms score in confirmed cases (con- stitutional illness) | 1 | 20 | Mean Difference (IV, Random, 95% CI) | -2.0 [-16.98, 12.98] |
| 3 Adverse effects - local - nasal irritation | 1 | 20 | Odds Ratio (M-H, Random, 95% CI) | 12.50 [1.09, 143.43] |

Amantadine and rimantadine for influenza A in adults (Review)

Copyright $\ensuremath{\textcircled{O}}$ 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.



| Outcome or subgroup title | No. of studies | No. of partici- pants | Statistical method | Effect size |
|--|-------------------|-----------------------------|----------------------------------|-------------------|
| 4 Viral nasal shedding or persistence in up- per airways at 2 to 5 days | 1 | 20 | Risk Ratio (M-H, Random, 95% CI) | 1.53 [0.58, 4.05] |

Analysis 9.1. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 1 Symptoms score in confirmed cases (respiratory illness).

| Study or subgroup | Inhaled amantadine | | Inhal | Inhaled placebo | | Mean Difference | | | | Weight | Mean Difference |
|--|--------------------|----------|---------|-----------------|-----|-----------------|-----------|----|----|----------------|-----------------|
| | Ν | Mean(SD) | Ν | Mean(SD) | | Ra | ndom, 95% | CI | | | Random, 95% Cl |
| Hayden 1980 | 9 | 8 (3) | 11 | 9 (3) | | | | | | 100% | -1[-3.64,1.64] |
| Total *** | 9 | | 11 | | | | | | | 100% | -1[-3.64,1.64] |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=0.74(P=0.46 |) | | | | | | | | | | |
| | | | Inhalor | lamantadine | -10 | -5 | 0 | 5 | 10 | Inhaled placeb | 2 |

Inhaled amantadine Inhaled placebo

Analysis 9.2. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 2 Symptoms score in confirmed cases (constitutional illness).

| Study or subgroup | Inhaled | amantadine | Inhaled placebo | | | Mean Difference | | | | Weight | Mean Difference |
|---------------------------------------|---------|------------|-----------------|------------|-----|-----------------|----------|------|----|-----------------|------------------|
| | Ν | Mean(SD) | Ν | Mean(SD) | | Rane | dom, 95% | b CI | | | Random, 95% Cl |
| Hayden 1980 | 9 | 12 (17) | 11 | 14 (17) | • | | | | | 100% | -2[-16.98,12.98] |
| Total *** | 9 | | 11 | | | | | | | 100% | -2[-16.98,12.98] |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=0.26(P=0.7 | 9) | | | | | | | | | | |
| | | | Inhaled | amantadine | -10 | -5 | 0 | 5 | 10 | Inhaled placebo |) |

Analysis 9.3. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 3 Adverse effects - local - nasal irritation.

| Study or subgroup | Inhaled amantadine | Inhaled placebo | | | Ode | ds Rat | io | | | Weight | Odds Ratio |
|-------------------------------------|----------------------------|--------------------|-----|-----|---------|--------|--------|---|----|-----------------|---------------------|
| | n/N | n/N | | M | -H, Ran | ndom, | 95% CI | | | | M-H, Random, 95% Cl |
| Hayden 1980 | 5/9 | 1/11 | | | | - | | | - | 100% | 12.5[1.09,143.43] |
| Total (95% CI) | 9 | 11 | | | | - | | | | 100% | 12.5[1.09,143.43] |
| Total events: 5 (Inhaled amantad | line), 1 (Inhaled placebo) | | | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=2.03(P=0 | 0.04) | | | | | | | | | | |
| | Inha | ed amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Inhaled placebo | |

Analysis 9.4. Comparison 9 Inhaled amantadine versus placebo (treatment), Outcome 4 Viral nasal shedding or persistence in upper airways at 2 to 5 days.

| Study or subgroup | Amantadine | Control | | | Ri | sk Rat | io | | | Weight | Risk Ratio |
|---|------------|-----------------|-----|-----|---------|--------|----------|---|----|-----------------|---------------------|
| | n/N | n/N | | | M-H, Ra | ndom | , 95% CI | | | | M-H, Random, 95% CI |
| Hayden 1980 | 5/9 | 4/11 | | | _ | | + | _ | | 100% | 1.53[0.58,4.05] |
| Total (95% CI) | 9 | 11 | | | - | | | - | | 100% | 1.53[0.58,4.05] |
| Total events: 5 (Amantadine), 4 (Contro | l) | | | | | | | | | | |
| Heterogeneity: Not applicable | | | | | | | | | | | |
| Test for overall effect: Z=0.85(P=0.39) | | | | | | | | | | | |
| | Fave | ours amantadine | 0.1 | 0.2 | 0.5 | 1 | 2 | 5 | 10 | Favours control | |

ADDITIONAL TABLES

Table 1. Trial symptom outcomes used

| Galbraith 1971 | Average time to clearance of symptoms |
|----------------|--|
| Hayden 1980 | Aggregate scores of systemic and respiratory symptoms |
| Hayden 1986 | Aggregate scores of systemic and respiratory symptoms |
| Hornick 1969a | Percentage of patients in three symptoms clearance time periods |
| Kitamoto 1968 | No symptoms |
| Kitamoto 1971 | No symptoms |
| Knight 1970 | Between arms symptoms concordance. Aggregate data only |
| Máté 1970 | Duration of fever (aggregate) and length of stay in infirmary |
| Togo 1970* | Percentage of patients in three symptoms clearance time periods |
| Younkin 1983* | Significance of the difference of symptoms scores |
| van Voris 1981 | Percentage of improvement of symptom scores at different time periods |
| Wingfield 1969 | Significance of difference of proportions of patients in three symptoms clearance time periods |
| | |

FEEDBACK

Missing study?

Summary

I recently stuck upon a paper that may be relevant for this review, either to include (am not sure whether the trial was randomized, but it was placebo-controlled) or as 'excluded study'.

Here are the details:

Máté J, Simon M, Juvancz I, et al. Prophylactic use of amantadine during Hong Kong influenza epidemic. Acta Microbiol Acad Sci Hung 1970;17: 285-296. Can send a copy if necessary.

I certify that I have no affiliations with or involvement in any organisation or entity with a direct financial interest in the subject matter of my criticisms



Reply

The study identified by Van Wouden has been included in the review.

Tom Jefferson

Contributors

Johannes C van der Wouden Feedback comment added 27/05/04

WHAT'S NEW

| Date | Event | Description |
|-------------|---------------------------|--|
| 7 June 2012 | Review declared as stable | Intervention superseded - as of 07 June 2012, this Cochrane Re- view is no longer being updated. The editorial team believes that the question addressed by this Cochrane Review no longer rel- evant to decision making, as amantadine and rimantadine for influenza A in adults has been replaced by neuraminidase in- hibitors and are no longer used. |

HISTORY

Protocol first published: Issue 3, 1998 Review first published: Issue 2, 1999

| Date | Event | Description |
|-------------------|--|---------------------------------|
| 30 April 2008 | Amended | Converted to new review format. |
| 26 April 2008 | New search has been performed | Searches conducted. |
| 14 February 2006 | New citation required and conclusions have changed | Substantive amendment. |
| 15 September 2005 | New search has been performed | Searches conducted. |
| 26 April 2004 | Feedback has been incorporated | Feedback and reply added. |
| 9 November 2003 | New search has been performed | Searches conducted. |
| 29 March 2001 | New search has been performed | Searches conducted. |
| 29 April 1998 | New search has been performed | Searches conducted. |

CONTRIBUTIONS OF AUTHORS

For the 2004 update, TOJ and VD wrote the protocol for the review. TOJ, VD and JJD (Jon Deeks) extracted the data, constructed the comparisons and drew conclusions. TOJ carried out the 2001 and 2004 updates. VD and JJD checked the text of the 2004 update. JJD advised on the inclusion of the Mate study.

For the 2005 update, TOJ and DR applied the inclusion criteria and extracted data while VD supervised extraction and arbitrated when necessary. CDP checked and transformed data and supervised the revised meta-analysis. TOJ edited the text and all authors contributed.

For this 2008 update, AR carried out the searches and TOJ screened the results.

DECLARATIONS OF INTEREST

None known.

SOURCES OF SUPPORT

Internal sources

- Ministry of Defence, UK.
- Health Reviews Ltd, UK.
- Regione Piemonte, Italy.

External sources

• No sources of support supplied

NOTES

In the 2004 update we included two more studies which had not been identified during the original searches (Máté 1970 and Nafta 1970) and updated text and references. We also assessed and excluded five more trials (Clover 1986; Dawkins 1968; Finklea 1967; Knight 1969; Togo 1968) and are awaiting translation of a further study from Polish (Tkaczewski 1972). The results and conclusions of the review do not change much. The confidence intervals around the effects of amantadine are narrower, but the findings on rimantadine are identical as both Máté and Nafta assessed the effects of amantadine.

The terms 'laboratory-confirmed influenza' and 'clinically confirmed influenza' have been changed for the more correct terms 'influenza' and 'influenza-like-illness' (ILI). We believe these words to reflect the difference between real influenza, caused by A and B viruses and what is colloquially known as 'the flu'. The two are rarely clinically distinguishable in real-time unless a very good surveillance apparatus is in place, as in most of the trials in our review.

The practical importance of this difference, which is rarely explained to the public, can be seen in the markedly different effectiveness profiles of the two drugs. Rarely can amantadine or rimantadine be used to good effect against ILI, which is what presents to both patient and doctor.

In the 2006 update we included one more treatment trial comparing rimantadine with placebo (Rabinovich 1969) and one comparing amantadine with standard treatment (Ito 2000) and excluded one more study, by Bricaire and colleagues. We also updated and shortened the text.

Because of the threat of a pandemic and on the basis of a comment made by Professor Robert B Couch we assessed the effectiveness of both compounds in preventing infection (as opposite to preventing or treating its symptoms) and in interrupting the chain of transmission (measured by the quantity and duration of viruses voided from the upper airways of infected people). We found no evidence of effectiveness of either compound. This led us to revise 'downwards' our estimates of effectiveness and warn readers that amantadine and rimantadine should be used only in emergencies. This conclusion was also based on the mounting evidence of resistance of influenza A viruses to both compounds.

In this April 2008 update we re-ran the searches but found no items relevant to the review. The conclusions and the text stand unaltered.

INDEX TERMS

Medical Subject Headings (MeSH)

*Influenza A virus; Amantadine [adverse effects] [*therapeutic use]; Antiviral Agents [adverse effects] [*therapeutic use]; Drug Administration Schedule; Emergencies; Influenza, Human [*drug therapy] [*prevention & control]; Randomized Controlled Trials as Topic; Rimantadine [adverse effects] [*therapeutic use]; Virus Shedding [drug effects]

MeSH check words

Adult; Aged; Humans; Middle Aged